## COMPOSITIO MATHEMATICA

## IAN M. ABERBACH CRAIG HUNEKE NGÔ VIÊT TRUNG Reduction numbers, Briançon-Skoda theorems and the depth of Rees rings

*Compositio Mathematica*, tome 97, nº 3 (1995), p. 403-434 <a href="http://www.numdam.org/item?id=CM">http://www.numdam.org/item?id=CM</a> 1995 97 3 403 0>

© Foundation Compositio Mathematica, 1995, tous droits réservés.

L'accès aux archives de la revue « Compositio Mathematica » (http: //http://www.compositio.nl/) implique l'accord avec les conditions générales d'utilisation (http://www.numdam.org/conditions). Toute utilisation commerciale ou impression systématique est constitutive d'une infraction pénale. Toute copie ou impression de ce fichier doit contenir la présente mention de copyright.

### $\mathcal{N}$ umdam

Article numérisé dans le cadre du programme Numérisation de documents anciens mathématiques http://www.numdam.org/

# Reduction numbers, Briançon–Skoda theorems and the depth of Rees rings

IAN. M. ABERBACH<sup>1</sup>, CRAIG HUNEKE<sup>2</sup> and NGÔ VIÊT TRUNG<sup>3</sup>\* <sup>1</sup>Department of Mathematics, 1409 W. Green St., University of Illinois, Urbana, IL 61801 Current address: Department of Mathematics, University of Missouri Columbia, Missouri 65211 E-mail address: aberbach@symcom.math.uiuc.edu <sup>2</sup>Department of Mathematics, Purdue University, West Lafayette, IN 47907 E-mail address: huneke@math.purdue.edu

<sup>3</sup>Institute of Mathematics, Box 631, Bò Hô, Hanoi, Vietnam

Received 6 December 1993; accepted in final form 5 May 1994

#### 1. Introduction

Let R be a commutative noetherian ring, and let  $I \subseteq R$  be an ideal of R. Over the last fifteen years, there has been considerable effort in the study of the Rees algebra of  $I, R[It] = \bigoplus_{n \ge 0} I^n$ , and the associated graded ring of  $I, G(I) = \bigoplus_{n \ge 0} I^n / I^{n+1}$ . The associated Proj of these two graded rings are respectively the blowup of I and the normal cone of I. One of the main questions addressed has been when these algebras are Cohen-Macaulay (which we will abbreviate CM). If they are, there is a wealth of information about how I sits in R. See the introduction of [HH2] for a discussion of this topic, and our bibliography for many related papers. Recently, due to work of Sancho de Salas [SS] and Joseph Lipman [L], there is even more reason to study the Cohen–Macaulayness of these algebras. For instance, when Ris CM with a canonical module, Sancho de Salas proves that  $G(I^n)$  is CM for large  $n \text{ iff } X = \operatorname{Proj}(R[It])$  is CM and satisfies a Grauert-Riemenschneider vanishing property, namely that  $R^i f_*(\omega_X) = 0$  for  $i \ge 1$ , where  $\omega_X$  is the dualizing sheaf of X. Lipman proves that provided R is essentially of finite type over the complex numbers and X is nonsingular, then R has rational singularities iff  $R[I^n t]$  is CM for large n.

A subsidiary question is the relationship between the CM property of R[It]and the CM property of G(I). In [Hul], it was shown that if R is CM, then R[It]CM implies G(I) CM. The converse need not hold. (Indeed, putting together the results of Sancho de Salas and Lipman, one obtains, at least for rings essentially of finite type over the complex numbers, that if the converse holds for every ideal

<sup>\*</sup> During the preparation of this paper the first author was supported in part by an NSF Mathematical Sciences Postdoctoral Fellowship, the second author was partially supported by the NSF, and the third author was partially supported by the National Basic Research Program, Vietnam, and the Max–Planck-Institute für Mathematik, Bonn.

then the ring has a rational singularity. On the other hand, while this paper was being written, Lipman has shown that if R is pseudo-rational then for every ideal Iof R, G(I) CM implies that R[It] is CM [L].) The first theorem giving necessary and sufficient conditions for the converse to hold in case I is m-primary was due to Goto and Shimoda [GS] in 1979. They use the concept of a reduction number. A reduction of I is an ideal  $J \subseteq I$  for which there exists an integer n such that  $JI^n = I^{n+1}$ . The least integer n with this property is called the *reduction number* of I with respect to J and we will denote it by  $r_J(I)$ . The reduction number of Iis defined by

 $r(I) := \min\{r_J(I)|J \text{ is a minimal reduction of } I\},\$ 

where J is said to be a *minimal reduction* of I if it is not properly contained in any other reduction of I. The theorem of Goto and Shimoda states:

THEOREM 1.1. Let (R, m) be a d-dimensional CM local ring with infinite residue field and let I be an m-primary ideal of R. Then R[It] is CM iff G(I) is CM and  $r(I) \leq d - 1$ .

This theorem was recently improved by Goto and Huckaba [GH], and related results have been given in [HM]. Trung and Ikeda gave a fairly definitive answer using the *a*-invariant of the graded ring: they proved in [TI] that if G(I) is CM, then R[It] is CM iff the *a*-invariant of G(I) is negative (see also [V] and [GN]). Other authors have recently studied similar questions for the Gorenstein property, cf. [GNi], [HHR], [HRS], [I] for example. In this paper we will give a full generalization of the theorem of Goto and Shimoda. See Theorem 5.1. While we were writing this paper, we learned that independently Bernie Johnston and Dan Katz found this result (cf. [JK]) and Aron Simis, Bernd Ulrich, and Wolmer Vasconcelos found closely related results ([SUV]). A recent paper by Tang ([Ta]) contains some related results. We also will give necessary and sufficient conditions for the Gorenstein property to hold for the Rees and graded algebras in Theorem 5.8. Moreover our method of proof yields considerable information concerning the regularity of the Rees algebra and what we call the local reduction numbers of an ideal.

The exact statements of Theorems 5.1 and 5.8 are:

THEOREM 5.1. Let (R, m) be a CM ring and I an ideal of R with  $ht(I) \ge 1$ . Then R[It] is CM if and only if the following conditions are satisfied:

(i) G(I) is Cohen–Macaulay.

(ii)  $r(I_{\wp}) \leq ht(\wp) - 1$  for every prime  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp)$ .

THEOREM 5.8. Let R be a Cohen–Macaulay ring and I an ideal of R with  $ht(I) \ge 2$ . Then R[It] is Gorenstein if and only if the following conditions are satisfied:

- (i) G(I) is Gorenstein.
- (ii)  $r(I_{\wp}) = ht(\wp) 2$  for every prime ideal  $\wp \supseteq I$  with  $ht(\wp/I) = 0$ .

(iii)  $r(I_{\wp}) \leq ht(\wp) - 2$  for every prime ideal  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp) \leq \ell(I)$ .

Observe that applying (5.1) in both directions one obtains that if G(I) is CM, then R[It] is CM iff  $R_p[I_pt]$  is CM for every prime p such that  $\ell(I_p) = ht(p)$ . This is a very useful criterion.

If (R, m) is a local ring with infinite residue field R/m, every minimal reduction of I is minimally generated by

 $\ell(I) := \dim \bigoplus_{n \ge 0} I^n / m I^n$ 

elements, and  $\ell(I)$  is called the analytic spread of I. It is well-known that

 $\operatorname{ht}(I) \leqslant \ell(I) \leqslant \dim R.$ 

The analytic deviation of I is defined by

 $\mathrm{ad}(I) := \ell(I) - \mathrm{ht}(I).$ 

The relation type of I, N(I), is defined to be the maximal degree of a minimal homogeneous generator of the ideal ker  $R[T_1, \ldots, T_n] \rightarrow R[It]$ , where I is *n*-generated and  $T_j \mapsto i_j t$ .

We introduce the notion of local reduction numbers of a graded algebra S over a local ring. By means of this notion we give upper bounds for the regularity of minimal reductions and the reduction number of the ideal  $S_+$  generated by the elements of positive degree of S (cf. Theorem 3.2). These upper bounds can be used to estimate the *a*-invariant and the Castelnuovo–Mumford regularity of Cohen– Macaulay graded algebras and associated graded rings of ideals. As we mentioned above, this work allows us to derive general criteria for the Cohen–Macaulay and Gorenstein property of Rees algebras of ideals having arbitrary analytic deviation. We are also able to prove similar theorems relating the Serre property  $S_k$  of R[It] to the Serre property  $S_{k-1}$  of G(I) (see Theorem 6.8). We give numerous corollaries of our results when additional hypotheses on the ideal I are given. We also derive a bound on the relation type of an ideal I when both R[It] and R are CM rings. The precise statement is as follows:

COROLLARY 6.7. Let (R, m) be a CM ring with infinite residue field. If  $I \subseteq R$  is an ideal of analytic spread  $\ell$  and R[It] is CM then

 $N(I) \leq \max\{r_{\ell-1}(I), r(I) + 1\} \leq r_{\ell}(I) + 1.$ 

The number  $r_j(I)$  is the *j*th local reduction number (see the definition following Lemma 4.2).

As in the paper [AH], the bounds on reduction numbers of ideals have a great deal to do with theorems of Briançon–Skoda type (see [LT], [LS], and [AH]). The simplest version of the Briançon–Skoda theorem states:

THEOREM 1.2. Let (R, m) be a d-dimensional regular local ring. For every ideal I and every reduction J of I,  $I^d \subseteq J$ .

There are by now many versions giving more information. A basic problem in improving this result lies in understanding the coefficients in the theorem. By this we mean the following: let R be a regular local ring, and let I be an m-primary ideal. If R has an infinite residue field, then I has a reduction J generated by delements  $x_1, \ldots, x_d$ . Given an element  $u \in I^d$ , we may write

 $u = a_1 x_1 + \dots + a_d x_d.$ 

The "best" choice of the coefficients  $a_i$  is what we seek. Of course, the  $a_i$  are only defined up to trivial syzygies of the regular sequence  $x_1, \ldots, x_d$ . The best one could hope for would be to say there is a choice of the  $a_i$  in  $I^{d-1}$ ; this is exactly the case when the reduction number of I is at most d - 1, and this relates in turn to the Cohen-Macaulay property of the Rees algebra. In Theorem 7.6 we give a new Briançon-Skoda type theorem, valid for regular local rings containing a field, which gives very precise information about a choice of the coefficients  $a_i$ . The proof of Theorem 7.6 uses reduction to characteristic p, and in characteristic puses ideas coming from the theory of tight closure. This material is of independent interest, and does not use the results in the rest of this paper. We refer the reader to Theorem 7.6 for the exact statement, which is somewhat technical. Although this section may be read independently from the rest of the paper, in Section 8 we apply our new Briançon-Skoda theorem to give bounds on the reduction number of an ideal I in an equicharacteristic regular local ring when G(I) satisfies Serre's property  $S_{\ell}$ , where  $\ell = \ell(I)$ . These results strengthen Theorem 5.1 when the graded ring G(I) is not necessarily CM.

We now give a short summary of what is in each section. Section 2 contains basic definitions and material which we will need in the proofs of the main theorems in Section 5. In Section 2 we study standard graded algebras and develop graded versions of various local ideas such as analytic spread, reduction numbers, etc. Of particular note and importance in the proofs is the concept of a filter-regular sequence due to Trung. Section 3 contains the main theorems relating the local reduction numbers of a graded ring to its regularity and *a*-invariants. This section contains the heart of the proofs of the main results. In Section 4 we translate the local data coming from a local ring R and ideal I to the graded data of G(I)and R[It]. Section 5 gives the proofs of the main results connecting the Rees and graded algebras, and gives many corollaries of these results. Section 6 gives information concerning the Serre properties of the Rees and graded rings, bounds on the Castelnuovo-Mumford regularity of the Rees algebra, and bounds on the relation type of I in terms of the local reduction numbers when R and R[It] are CM. Section 7 gives our new Briançon-Skoda theorem, and Section 8 gives applications of it.

#### 2. Filter-regular sequences and minimal reductions of a graded algebra

In this section we collect some basic results on filter-regular sequences and minimal reductions of graded algebras which we shall need later in our study of the Cohen–Macaulay property of the associated graded ring and the Rees algebra.

By a graded algebra we always mean a graded algebra  $S = \bigoplus_{n \ge 0} S_n$  over a local ring  $S_0$  with infinite residue field such that  $S = S_0[S_1]$  (standard graded algebra). We denote by M the maximal graded ideal of S and by  $S_+ = \bigoplus_{n>0} S_n$  the ideal generated by all homogeneous elements of positive degree of S. As usual, for any graded module L over S, and any integer n,  $L_n$  denotes the n-graded part of L.

A system of homogeneous elements  $z_1, \ldots, z_r$  of S is called a *filter-regular* sequence on S (with respect to  $S_+$ ) if  $z_i \notin P$  for any prime  $P \in Ass(S/(z_1, \ldots, z_{i-1}))$ ,  $P \not\supseteq S_+$ ,  $i = 1, \ldots, r$ . This notion originated from the theory of generalized Cohen-Macaulay rings [CST] where filter-regular sequences are defined for local rings with respect to the maximal ideal. We refer the reader to [CST] and [T2] for basic properties of filter-regular sequences.

Given any ideal Z of S we will put

 $\mathcal{U}(Z) := \bigcup_{n \ge 0} Z : (S_+)^n.$ 

For any primary decomposition of Z, U(Z) is the intersection of the primary components of Z whose associated primes do not contain  $S_+$ . A sequence  $z_1, \ldots, z_r$  of homogeneous elements in S is a filter-regular sequence on S if and only if

 $(z_1,\ldots,z_{i-1}):z_i\subseteq\mathcal{U}((z_1,\ldots,z_{i-1})),$ 

for i = 1, ..., r (see e.g. [T2]).

LEMMA 2.1. Let  $z_1, \ldots, z_r$  be a filter-regular sequence on S. Then

(i)  $\mathcal{U}((z_1, \ldots, z_{r-1})) : z_r = \mathcal{U}((z_1, \ldots, z_{r-1})).$ 

(ii)  $ht(P) \ge r$  for every associated prime P of  $\mathcal{U}(z_1, \ldots, z_r)$ .

*Proof.* Since every associated prime P of  $\mathcal{U}((z_1, \ldots, z_{r-1}))$  does not contain  $S_+$ , we have that  $z_r \notin P$  by the definition of filter-regular sequences. From this (i) follows. For (ii) we first note that since  $P \not\supseteq S_+$ ,  $z_i \notin Q$  for any prime  $Q \in \operatorname{Ass}(S/(z_1, \ldots, z_{i-1})), Q \subseteq P, i = 1, \ldots, r$ . Therefore,  $z_1, \ldots, z_r$  is a regular sequence of  $S_P$ . It follows that  $\dim(S_P) \ge r$ . Since  $\operatorname{ht}(P) = \dim(S_P)$ , this gives the conclusion.  $\Box$ 

COROLLARY 2.2. Let  $z_1, \ldots, z_d \in S_+$  be a filter-regular sequence on S with respect to  $S_+$ ,  $d = \dim S$ . Then  $\sqrt{S_+} = \sqrt{(z_1, \ldots, z_d)}$ .

*Proof.* Set  $Z = (z_1, \ldots, z_d)$ . By Lemma 2.1 (ii) we have  $\mathcal{U}(Z) = S$ . Hence  $(S_+)^n \subset Z$  for n sufficiently large. Since  $Z \subseteq S_+$ , the conclusion is immediate.  $\Box$ 

LEMMA 2.3. Let S be a Cohen–Macaulay graded algebra. Then any filter-regular sequence  $z_1, \ldots, z_s$  on S with length  $s = ht(S_+)$  is a regular sequence.

*Proof.* We will prove by induction that  $z_1, \ldots, z_i$  is a regular sequence for  $i = 0, \ldots, s$ . For i = 0 there is nothing to prove. For i > 0, we may assume that  $z_1, \ldots, z_{i-1}$  is a regular sequence of S. Let  $P \in \operatorname{Ass}(S/(z_1, \ldots, z_{i-1}))$ . Since S is Cohen-Macaulay,  $\operatorname{ht}(P) = i - 1$ . Since  $\operatorname{ht}(S_+) = s > i - 1$ ,  $P \not\supseteq S_+$ . By the definition of filter-regular sequences,  $z_i \notin P$ . Hence  $z_1, \ldots, z_i$  is also a regular sequence on S.

A filter-regular sequence  $z_1, \ldots, z_r$  of S can also be characterized by its regularity. Following [T1] and [AH] we say that a sequence  $z_1, \ldots, z_r$  of homogeneous elements of S is  $[t_1, \ldots, t_r]$ -regular if

$$[(z_1,\ldots,z_{i-1}):z_i]_n=(z_1,\ldots,z_{i-1})_n,$$

for  $n \ge t_i$ , i = 1, ..., r. The value of  $t_i$  is also allowed to be  $-\infty$  and  $+\infty$ . According to [T1, Lemma 2.1]  $z_1, ..., z_r$  is filter-regular if and only if there exist integers  $t_i < +\infty$  such that  $z_1, ..., z_r$  is  $[t_1, ..., t_r]$ -regular.

Now we will establish some properties of minimal reductions of the ideal  $S_+$ .

LEMMA 2.4. An ideal Z generated by 1-forms is a reduction of  $S_+$  if and only if  $\sqrt{S_+} = \sqrt{Z}$ . In this case,

 $r_Z(S_+) = \min\{n | S_{n+1} = Z_{n+1}\},\$ 

The proof is left to the reader.

As in the local case [NR] one can show that a minimal reduction of  $S_+$  is minimally generated by

$$\ell(S_+) := \dim\left(\bigoplus_{n \ge 0} S_n / \mathfrak{m} S_n\right)$$

elements, where m denotes the maximal ideal of  $S_0$ . We call  $\ell(S_+)$  the analytic spread of  $S_+$ . By [T1, Lemma 3.1] every minimal reduction of  $S_+$  can be minimally generated by a filter-regular sequence of S. Hence we can introduce the following definitions.

DEFINITION 2.5. Let Z be a minimal reduction of  $S_+$ . We call the least number t such that there exists a minimal generating set  $z_1, \ldots, z_\ell$  of Z which is  $[t, \ldots, t]$ -regular (resp.  $[t + 1, \ldots, t + \ell]$ -regular) the *regularity* (resp. the *sliding regularity*) of Z. This number will be denoted by a(Z) (resp. s(Z)).

The numbers s(Z), a(Z) and  $\ell(S_+)$  play a crucial role in the estimation of the *a*-invariant and the Castelnuovo–Mumford regularity of S. Recall that the *a*-invariant of S [GW] is defined by

$$a(S) := \max\{n \mid [H_M^d(S)]_n \neq 0\},\$$

where  $d := \dim S$  and  $H^d_M(S)$  denotes the *d*th local cohomology module of *S* with respect to the maximal graded ideal *M*. The Castelnuovo–Mumford regularity of *S* [Mu] is defined by

$$\operatorname{reg}(S) := \max_{i \ge 0} \{ \min\{t \mid [H^i_{S_+}(S)]_{n-i} = 0 \text{ for } n \ge t \} + i \} - 1,$$

where  $H_{S_{+}}^{i}(S)$  denotes the *i*th local cohomology module of S with respect to  $S_{+}$ .

A special case of [T3, Theorem 2.2] and [T3, Corollary 2.3] is

THEOREM 2.6. Let S be a Cohen–Macaulay graded algebra and let Z be a minimal reduction of  $S_+$ . Then  $a(S) \leq s(Z)$  if and only if  $r_Z(S_+) \leq \ell(S_+) + s(Z)$ . In addition, for any b > s(Z), a(S) = b if and only if  $r(S_+) = \ell(S_+) + b$ .

**PROPOSITION 2.7.** Let S be a Cohen–Macaulay graded algebra. Let Z be a minimal reduction of  $S_+$ . Then

$$\max\{s(Z), a(S)\} = \max\{s(Z), r_Z(S_+) - \ell(S_+)\}.$$

*Proof.* If  $a(S) \leq s(Z)$  then  $r_Z(S_+) - \ell(S_+) \leq s(Z)$  by Theorem 2.6 so both sides are equal to s(Z). Suppose a(S) > s(Z). Then by Theorem 2.6,  $r_Z(S_+) - \ell(S_+) = a(S)$  and the equality again holds.  $\Box$ 

REMARK 2.8. One can divide the above relation into two parts:

- (1)  $r_Z(S_+) \leq \max\{s(Z), a(S)\} + \ell(S_+).$
- (2)  $a(S) \leq \max\{s(Z), r_Z(S_+) \ell(S_+)\}.$

COROLLARY 2.9. Let S be a Cohen-Macaulay graded algebra. Let Z be a minimal reduction of  $S_+$ . If Y is any minimal reduction of  $S_+$  such that  $r(S_+) = r_Y(S_+)$  then

 $r_Z(S_+) \leqslant \max\{s(Z) + \ell(S_+), r(S_+), s(Y) + \ell(S_+)\}.$ 

*Proof.* By applying Proposition 2.7 with both Z and Y we have  $r_Z(S_+) \leq \max\{s(Z), a(S)\} + \ell(S_+) \leq \max\{s(Z), s(Y), r_Y(S_+) - \ell(S_+)\} + \ell(S_+)$  since  $a(S) \leq \max\{s(Y), r_Y(S_+) - \ell(S_+)\}$ .

**PROPOSITION 2.10.** [T1, Corollary 3.3]. Let S be a graded algebra and Z a minimal reduction of  $S_+$ . Then

 $\operatorname{reg}(S) \leq \max\{a(Z) - 1, r_Z(S_+)\}.$ 

REMARK 2.11. Mumford [Mu] showed that reg(S) gives upper bounds for the degrees of the generators and of all higher order syzygies of the defining ideal of S in a polynomial ring over  $S_0$ . In particular, the maximum degree of the elements of a homogeneous minimal basis of the defining ideal of S is bounded above by reg(S) + 1 (see also [EG], [O]).

#### 3. Regularity and local reduction numbers of a graded algebra

Let  $S = \bigoplus_{n \ge 0} S_n$  be a (standard) graded algebra over a local ring  $S_0$  with infinite residue field. As we have seen in Propositions 2.7 and 2.10 the regularities and the reduction numbers of minimal reductions of  $S_+$  provide information on the *a*-invariant and the Castelnuovo–Mumford regularity of *S*. In this section we will estimate these invariants in terms of simpler invariants of *S*.

Given a prime ideal  $\mathfrak{p}$  of  $S_0$  we will denote by  $S_{\mathfrak{p}}$  the localization of S at the multiplicatively closed set  $S_0 \setminus \mathfrak{p}$ .  $S_{\mathfrak{p}} = \bigoplus_{n \ge 0} (S_n)_{\mathfrak{p}}$  is a graded algebra over the local ring  $(S_0)_{\mathfrak{p}}$ . If  $P = \mathfrak{p}S + S_+$  where  $\mathfrak{p} \in \operatorname{Spec} S_0$  then dim  $S_{\mathfrak{p}} = \dim S_P = \operatorname{ht} P$ . Set  $s = \operatorname{ht}(S_+)$ ,  $\ell = \ell(S_+)$ . We introduce the following invariants:

$$r_{i}(S) := \begin{cases} i-s, & i < s, \\ \max\{r((S_{\mathfrak{p}})_{+}) - \operatorname{ht}(\mathfrak{p}) \mid \ell((S_{\mathfrak{p}})_{+}) = \operatorname{ht}(\mathfrak{p}) + s \leqslant i\} + i-s, \ s \leqslant i \leqslant \ell \end{cases}$$

which we call the *ith local reduction number* of S. (Note that we allow i to take negative values.)

REMARK 3.1.

(1) For  $i = s, ..., \ell$  we have the following relation between  $r_{i-1}(S)$  and  $r_i(S)$ :

$$r_i(S) = \max\{r_{i-1}(S) + 1, r((S_{\mathfrak{p}})_{+}) | \ell((S_{\mathfrak{p}})_{+}) = \operatorname{ht}(\mathfrak{p}) + s = i\}.$$

- (2)  $r_0(S) < \cdots < r_{\ell}(S)$  is a strictly increasing sequence of numbers.
- (3) If  $\mathfrak{p} \in \text{Spec } S_0$  then  $r_i(S_\mathfrak{p}) \leq r_i(S)$  for  $i \leq \ell((S_\mathfrak{p})_+)$ .

THEOREM 3.2. Let S be a Cohen–Macaulay graded algebra. Set  $\ell = \ell(S_+)$  and let  $Z = (z_1, \ldots, z_\ell)$  be a minimal reduction of S. Then

(i)  $z_1, \ldots, z_\ell$  is  $[r_0(S) + 1, \ldots, r_{\ell-1}(S) + 1]$ -regular for any filter-regular sequence  $z_1, \ldots, z_\ell$  of S which generates Z.

(ii) 
$$r_Z(S_+) \leqslant r_\ell(S)$$
.

We will prove Theorem 3.2 by proving:

$$(C_i): [(z_1, \dots, z_i): z_{i+1}]_n = (z_i, \dots, z_i)_n \text{ for } n \ge r_i(S) + 1 \ 0 \le i < \ell.$$
  
$$(C_\ell): r_Z(S_+) \le r_\ell(S).$$

For each *i* such that  $0 \le i \le \ell$ , fix a primary decomposition of the ideal  $(z_1, \ldots, z_i)$  and let  $U_i$  be the intersection of the primary components which contain  $S_+$  and have height at most *i*. Also, let  $V_i = \mathcal{U}((z_1, \ldots, z_i))$ .

For the proof of Theorem 3.2 we need the following lemmas.

LEMMA 3.3. Let  $U \subseteq L$  be two graded ideals of S. For an integer  $n \ge 0$ ,  $U_n = L_n$  if and only if  $[U_p]_n = [L_p]_n$  for every prime p of  $S_0$  which is the contraction of an associated prime of U.

*Proof.*  $(\Rightarrow)$  is trivial. To see  $(\Leftarrow)$  we have to show that  $L_n \subseteq U_n$ . For this it suffices to show that  $L_n \subseteq Q_n$  for every primary component Q of U. Let  $\mathfrak{p}$  be the contraction in  $S_0$  of the associated prime of Q. Then  $[L_n]_{\mathfrak{p}} = [L_{\mathfrak{p}}]_n = [U_{\mathfrak{p}}]_n \subseteq [Q_n]_{\mathfrak{p}}$ . From this is follows that  $L_n \subseteq Q_n$ .

LEMMA 3.4. Let S be a CM graded algebra with  $s = ht(S_+)$  and  $\ell = \ell(S)$ . Let  $Z = (z_1, \ldots, z_\ell)$  be a minimal reduction of  $S_+$ . Fix i such that  $s \leq i \leq \ell$ . Assume that the sequence  $z_1, \ldots, z_\ell$  satisfies  $(C_j)$  for  $0 \leq j < i$ . Let  $P \supseteq S_+$  be a prime ideal of S with htP > i. Let p be the contraction of P to  $S_0$ . Then

 $[H_{P_{\mathfrak{p}}}^{k}(S_{\mathfrak{p}}/(z_{1},\ldots,z_{i})_{\mathfrak{p}})]_{n}=0,$ 

for  $n \ge r_{i-1}(S) + 2$ , k < ht(P) - i.

*Proof.* We show the result by induction. Let  $r_j = r_j(S)$  for  $0 \le j \le \ell$ . Notice that  $P = \mathfrak{p}S + S_+$ . The case i = s follows from the fact that  $S/(z_1, \ldots, z_s)$  is a Cohen-Macaulay ring. For, since  $S_{\mathfrak{p}}/(z_1, \ldots, z_s)_{\mathfrak{p}}$  is also a Cohen-Macaulay ring with dimension  $\operatorname{ht}(P) - s$ 

$$H_{P_{\mathfrak{p}}}^{k}(S_{\mathfrak{p}}/(z_{1},\ldots,z_{s})_{\mathfrak{p}})=0,$$

for  $k < \operatorname{ht}(P) - s$ .

If  $j \ge s + 1$ , we consider the exact sequence

$$0 \to [(z_1, \dots, z_{j-1}) : z_j]/(z_1, \dots, z_{j-1}) \to S_+/(z_1, \dots, z_{j-1}) \\ \to S/[(z_1, \dots, z_{j-1}) : z_j] \to 0.$$

Localized at p this induces the following exact sequence of local cohomology modules

$$[H_{P_{\mathfrak{p}}}^{k}(S_{\mathfrak{p}}/(z_{1},\ldots,z_{j-1})_{\mathfrak{p}})]_{n} \to [H_{P_{\mathfrak{p}}}^{k}(S_{\mathfrak{p}}/((z_{1},\ldots,z_{j-1}):z_{j})_{\mathfrak{p}})]_{n} \to [H_{P_{\mathfrak{p}}}^{k+1}(([(z_{1},\ldots,z_{j-1}):z_{j}]/(z_{1},\ldots,z_{j-1}))_{\mathfrak{p}})]_{n}$$

By the inductive hypothesis we have

 $[H_{P_{\mathfrak{p}}}^{k}(S_{\mathfrak{p}}/(z_{1},\ldots,z_{j-1})_{\mathfrak{p}})]_{n}=0,$ 

for  $n \ge r_{j-2} + 2$ , k < ht(P) - j + 1 (this holds even if j = 1 and s = 0). Further,  $(C_{j-1})$  is satisfied. Hence

$$[H_{P_{\mathfrak{p}}}^{m}([(z_{1},\ldots,z_{j-1})_{\mathfrak{p}}:z_{j}]/(z_{1},\ldots,z_{j-1})_{\mathfrak{p}})]_{n}=0,$$

for  $n \ge r_{j-1} + 1$  and *m* arbitrary by Lemma 3.7 below. By Remark 3.1,  $r_{j-2} + 2 \le r_{j-1} + 1$ , so we can deduce that

$$[H_{P_{\mathfrak{p}}}^{k}(S_{\mathfrak{p}}/((z_{1},\ldots,z_{j-1}):z_{j})_{\mathfrak{p}})]_{n}=0,$$

for  $n \ge r_{j-1} + 1$ , k < ht(P) - j + 1. On the other hand, the exact sequence

$$0 \to S/[(z_1, \dots, z_{j-1}) : z_j](-1) \xrightarrow{z_j} S/(z_1, \dots, z_{j-1})$$
$$\longrightarrow S/(z_1, \dots, z_j) \to 0$$

yields the following exact sequence of local cohomology modules

$$[H_{P_{\mathfrak{p}}}^{k}(S_{\mathfrak{p}}/(z_{1},\ldots,z_{j-1})_{\mathfrak{p}})]_{n} \rightarrow [H_{P_{\mathfrak{p}}}^{k}(S_{\mathfrak{p}}/(z_{1},\ldots,z_{j})_{\mathfrak{p}})]_{n}$$
$$\rightarrow [H_{P_{\mathfrak{p}}}^{k+1}(S_{\mathfrak{p}}/((z_{1},\ldots,z_{j-1}):z_{i})_{\mathfrak{p}})]_{n-1}.$$

Hence we can conclude that  $[H_{P_{\mathfrak{p}}}^{k}(S_{\mathfrak{p}}/(z_{1},\ldots,z_{j})_{\mathfrak{p}})]_{n} = 0$ , for  $n \ge r_{j-1}+2$ , k < ht(P) - j.

LEMMA 3.5. Let S, Z and i be as in Lemma 3.4. Let  $d = \dim S$ . Assume that  $(C_j)$  holds for  $0 \leq j < i$ . Then  $(z_1, \ldots, z_i)_n = [U_i \cap V_i]_n$  for  $n \geq r_{i-1}(S) + 2$ .

**Proof.** Let  $r_j = r_j(S)$  for  $0 \le j \le \ell$ . For every prime ideal  $P \supseteq S_+$  of S let U(P), resp.  $U_0(P)$ , denote the intersection of the primary components of  $(z_1, \ldots, z_i)$  (from our fixed primary decomposition) whose associated primes are contained, resp. properly contained, in  $P(U(P) = S, \text{ resp. } U_0(P) = S$ , if there are no such primes). Let p be the contraction of P in  $S_0$ . Then

$$U_0(P)_{\mathfrak{p}}/U(P)_{\mathfrak{p}} = H^0_{P_{\mathfrak{p}}}(S_{\mathfrak{p}}/(z_1,\ldots,z_i)_{\mathfrak{p}})$$

This can be seen easily after localizing at  $\mathfrak{p}$ , so that  $\mathfrak{p} = m$  and  $P = m \oplus S_+$ . Note that then  $U(P) = (z_1, \ldots, z_i)$  so

$$U_0(P)/U(P) = \left[\bigcup_n (z_1, \dots, z_i) : P^n\right]/(z_1, \dots, z_i) = H_P^0(S/(z_1, \dots, z_i)).$$

If ht(P) > i, Lemma 3.4 applied with k = 0 implies that  $[U(P)_{\mathfrak{p}}]_n = [U_0(P)_{\mathfrak{p}}]_n$ for  $n \ge r_{i-1} + 2$ . Therefore  $[U(P)_{\mathfrak{q}}]_n = [U_0(P)_{\mathfrak{q}}]_n$  for any prime ideal  $\mathfrak{q} \subseteq \mathfrak{p}$  and  $n \ge r_{i-1} + 2$ . By Lemma 3.3, this means that

$$U(P)_n = U_0(P)_n,\tag{*}$$

for  $n \ge r_{i-1} + 2$ .

For every integer  $j \ge i$ , let  $W_j$  be the intersection of the (fixed) primary components of  $(z_1, \ldots, z_i)$  whose associated primes contain  $S_+$  and have height  $\le j$  ( $W_i = U_i$ ). It is easily seen that  $(z_1, \ldots, z_i) = W_d \cap V_i$  and

$$W_j \cap V_i = \bigcap_{P \supseteq S_+, \, \operatorname{ht}(P) = j} U(P) \cap V_i,$$

$$W_{j-1} \cap V_i = \bigcap_{P \supseteq S_+, \, \operatorname{ht}(P)=j} U_0(P) \cap V_i.$$

According to (\*),  $[W_j \cap V_i]_n = [W_{j-1} \cap V_i]_n$  for  $n \ge r_{i-1} + 2$ ,  $j = i + 1, \dots, d$ . Therefore we can conclude that  $(z_1, \dots, z_i)_n = [U_i \cap V_i]_n$  for  $n \ge r_{i-1} + 2$ .  $\Box$ 

Proof of Theorem 3.2. We will use induction on  $d = \dim(S)$ . If d = 0 then Z = 0 and  $s = \ell = 0$  so we need only to show  $(C_0)$ . This holds since  $r_0(S) = r(S_+)$  in this case.

Assume now that d > 0. We will prove by induction that  $(C_i)$  holds whenever i < d. If i < s then this is clear. Assume that  $i \ge s$  and assume by induction that  $(C_i)$  holds for  $0 \le j < i$ .

By Lemma 3.5 we know that  $(z_1, \ldots, z_i)_n = [U_i \cap V_i]_n$  for  $n \ge r_i(S) + 1$  since  $r_i(S) + 1 \ge r_{i-1}(S) + 2$ . Hence we must show that  $[U_i \cap V_i]_n : z_{i+1} = [U_i \cap V_i]_n$  for  $n \ge r_i(S) + 1$ . By Lemma 2.1(i),  $V_i : z_{i+1} = V_i$ . We claim that  $[U_i]_n = S_n$  for  $n \ge r_i(S) + 1$ . Let  $P \in Ass(S/U_i)$  and let p be the contraction of P to  $S_0$ . Since  $P \supseteq S_+$ , p is not the maximal ideal of  $S_0$ . Let  $k = \min\{i, \ell((S_p)_+)\}$ . Applying our induction hypothesis to  $S_p$  we know that  $r_{(z_1,\ldots,z_i)_p}((S_p)_+) \le r_k(S_p) \le r_i(S)$ . The last inequality follows from Remark 3.1 (3). This shows that  $[(U_i)_p]_n = [S_p]_n$  for  $n \ge r_i(S) + 1$ , so  $[U_i]_n = S_n$  for  $n \ge r_i(S) + 1$  by Lemma 3.3. Hence for  $n \ge r_i(S) + 1$ ,

$$\begin{aligned} [U_i \cap V_i]_n : z_{i+1} &= ([U_i]_n : z_{i+1}) \cap ([V_i]_n : z_{i+1}) \\ &= (S_n : z_{i+1}) \cap [V_i]_n = S_n \cap [V_i]_n = [U_i \cap V_i]_n, \end{aligned}$$

which shows  $(C_i)$  if  $i < \ell$ . Note that knowing  $(C_j)$  holds for  $j < \ell$  implies that  $s(Z) \leq r_{l-1}(S) - \ell + 1$ , and then Corollary 2.9 gives that

$$r_Z(S_+) \leq \max\{r(S_+), r_{\ell-1}(S) + 1\} \leq \max\{r(S_+), r_{\ell}(S)\},\$$

hence  $(C_{\ell})$  holds.

We obtain from Theorem 3.2 the following estimates for the regularities and the reduction number of minimal reductions of  $S_+$  in terms of the local reduction numbers.

COROLLARY 3.6. Let S be a Cohen–Macaulay graded algebra. Let Z be an arbitrary minimal reduction of  $S_+$ . Set  $\ell = \ell(S_+)$ . Then

- (i)  $s(Z) \leq r_{\ell-1}(S) \ell + 1$ .
- (ii)  $a(Z) \leq r_{\ell-1}(S) + 1$ .
- (iii)  $\max\{r_{\ell-1}(S) + 1, r_Z(S_+)\} = r_{\ell}(S).$

*Proof.* (i) and (ii) follow from Theorem 3.2(i) and Remark 3.1 (2). For (iii) we have  $r((S_{\mathfrak{p}})_+) \leq r(S_+)$  for every prime ideal  $\mathfrak{p}$  of  $S_0$  with  $\ell((S_{\mathfrak{p}})_+) = ht(\mathfrak{p}) + s = \ell$ 

because the image of a minimal reduction of  $S_+$  in  $S_p$  is also a minimal reduction of  $(S_p)_+$ . By the definition of  $r_i(S)$  this implies

$$r_{\ell}(S) \leq \max\{r_{\ell-1}(S)+1, r(S_{+})\} \leq \max\{r_{\ell-1}(S)+1, r_{Z}(S_{+})\}.$$

On the other hand, we have  $r_Z(S_+) \leq r_\ell(S)$  by Theorem 3.2 (ii). Since  $r_{\ell-1}(S) < r_\ell(S)$ , we get max $\{r_{\ell-1}(S)+1, r_Z(S_+)\} \leq r_\ell(S)$ .

LEMMA 3.7. Let S be an  $\mathbb{N}$ -graded Noetherian ring and  $I \subseteq R$  a homogeneous ideal of S. If M is a graded S-module such that  $M_n = 0$  for all  $n \ge j$  then  $[H_I^k(M)]_n = 0$  for all  $n \ge j$  and for all k.

*Proof.* Let  $f_1, \ldots, f_t$  be homogeneous generators of I. Then  $H_I^k(M)$  is a quotient of a submodule of the direct sum of the  $\mathbb{Z}$ -graded modules  $M_{f_{i_1} \cdots f_{i_k}}$ , where  $1 \leq i_1 < \cdots < i_k \leq t$ . Since deg  $f_i \geq 0$ , these modules vanish in degree j and higher.

#### 4. Local reduction numbers of an ideal

Throughout this section let (R, m) be a local Noetherian ring with infinite residue field. We will apply the results of Section 3 to the associated graded ring  $G(I) := \bigoplus_{n \ge 0} I^n / I^{n+1}$  of an ideal I in R. For this we need to make some observations on the relationship between certain invariants of  $G(I)_+$  and I.

LEMMA 4.1. The following relations hold for any ideal I in R:

- (i)  $\ell(G(I)_+) = \ell(I)$ .
- (ii)  $r_Z(G(I)_+) = r_J(I)$  whenever Z is generated by the initial forms of the elements of J in G(I).
- (iii)  $r(G(I)_+) = r(I)$ .

*Proof.* Set S = G(I). For (i) let m denote the maximal ideal m/I of the local ring  $R/I = S_0$ . It is clear that  $S_n/\mathfrak{m}S_n = I^n/mI^n$  for all  $n \ge 0$ . Therefore

$$\ell(S_+) = \dim\left(\bigoplus_{n \ge 0} S_n / \mathfrak{m} S_n\right) = \dim\left(\bigoplus_{n \ge 0} I^n / m I^n\right) = \ell(I).$$

For (ii) we note that  $r_Z(S_+)$  is the least integer *n* for which  $Z_{n+1} = S_{n+1}$ . This is equivalent to  $JI^n + I^{n+2} = I^{n+1}$  or, by Nakayama's Lemma,  $JI^n = I^{n+1}$  so that  $r_Z(S_+) = r_J(I)$ . Finally (iii) follows from (ii) because every minimal reduction Z of  $S_+$  is generated by the initial forms of the elements of a minimal reduction J of I and conversely.

Let L(I) denote the set of all prime ideals  $\wp \supseteq I$  of R with  $\ell(I_{\wp}) = ht(\wp)$ . This set is relatively small by the following lemma.

**LEMMA 4.2.** L(I) is a finite set and  $ht(\wp) \leq \ell(I)$  for every  $\wp \in L(I)$ .

**Proof.** L(I) is finite because L(I) is contained in the set  $\overline{A}^*(I) := \bigcup_{n \ge 1} \operatorname{Ass}(R/(\overline{I^n}))$ , where  $(\overline{I^n})$  denotes the integral closure of  $I^n$  [Mc, Proposition 4.1] and  $\overline{A}^*(I)$  is a finite set. For the second statement we note that every minimal reduction of an ideal is minimally generated by a number of elements which is equal to the analytic spread. Therefore, since the image of every minimal reduction of I in  $R_{\wp}$  is a reduction of  $I_{\wp}$ ,  $\ell(I) \ge \ell(I_{\wp}) = \operatorname{ht}(\wp)$ .

Put s = ht(I) and  $\ell = \ell(I)$ . We introduce the following invariants:

$$r_i(I) := \begin{cases} i - s, & i < s, \\ \max\{r(I_{\wp}) - \operatorname{ht}(\wp) | \wp \in L(I) \text{ and } \operatorname{ht}(\wp) \leqslant i\} + i, \ i = s, \dots, \ell, \end{cases}$$

which we call the *ith local reduction number of I*. Note that

$$r_s(I) = \max\{r(I_{\wp}) | \wp \supseteq I \text{ is a prime ideal of } R \text{ with } \operatorname{ht}(\wp) = s\}.$$

LEMMA 4.3. Assume that G(I) is equidimensional and catenary (for instance, this happens if R is universally catenary and equidimensional). Then

$$r_i(G(I)) = r_i(I),$$

for  $i = 0, ..., \ell(I)$ .

*Proof.* Set S = G(I). For every prime ideal  $\wp \supseteq I$  of R we denote by  $\mathfrak{p}$  its image in  $R/I = S_0$ . This gives a one-to-one correspondence between prime ideals of R which contain I and prime ideals of  $S_0$ . Since  $S_{\mathfrak{p}} = G(I_{\wp})$ , we have  $\ell(I_{\wp}) = \ell((S_{\mathfrak{p}})_+)$  and  $r(I_{\wp}) = r((S_{\mathfrak{p}})_+)$  by Lemma 4.1. If S is equidimensional and catenary, then  $ht(\wp) = ht(\mathfrak{p}) + s$ , hence

$$r((S_{\mathfrak{p}})_{+}) - \mathsf{ht}(\mathfrak{p}) + i - s = r(I_{\wp}) - \mathsf{ht}(\wp) + i.$$

By the definitions of the local reduction numbers of S and I, this implies  $r_i(S) = r_i(I)$  for  $i = 0, ..., \ell(I)$ .

THEOREM 4.4. Assume that G(I) is a Cohen–Macaulay ring. Set  $\ell = \ell(I)$  and let J be an arbitrary minimal reduction of I. Then

(i)  $\max\{r_{\ell-1}(I)+1, r_J(I)\} = r_{\ell}(I).$ 

(ii)  $\max\{r_{\ell-1}(I) + 1, a(G(I)) + \ell\} = r_{\ell}(I).$ 

*Proof.* Set S = G(I) and let Z be the minimal reduction of  $S_+$  generated by the initial forms of the elements of J in S. By Corollary 3.6(iii) and using Lemma 4.1 and Lemma 4.3 we have

$$\max\{r_{\ell-1}(I) + 1, r_J(I)\} = \max\{r_{\ell-1}(S) + 1, r_Z(S_+)\}\$$
  
=  $r_{\ell}(S) = r_{\ell}(I)$ 

I. M. ABERBACH ET AL.

which proves (i). For (ii) we have

 $\max\{s(Z) + \ell, a(S) + \ell\} = \max\{s(Z) + \ell, r_Z(S_+)\}$ 

by Proposition 2.7. By Corollary 3.6 (i),  $s(Z) + \ell \leq r_{\ell-1}(S) + 1$ . Hence

$$\max\{r_{\ell-1}(S)+1, a(S)+\ell\} = \max\{r_{\ell-1}(Z)+1, r_Z(S_+)\}.$$

Using Lemma 4.1 and Lemma 4.3 we can interpret this relation as

$$\max\{r_{\ell-1}(I)+1, a(S)+\ell\} = \max\{r_{\ell-1}(I)+1, r_J(I)\}\$$

which together with (i) proves (ii).

COROLLARY 4.5. Assume that G(I) is a Cohen–Macaulay ring. Set  $\ell = \ell(I)$ . If there exists a minimal reduction J of I such that  $r(I_{\wp}) - ht(\wp) < r_J(I) - \ell$  for every prime ideal  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp) < \ell$ , then

 $a(G(I)) = r_J(I) - \ell.$ 

*Proof.* The assumption on J means that  $r_J(I) > r_{\ell-1}(I) + 1$ . By Theorem 4.4 we get

$$r_J(I) = r_\ell(I) = \max\{r_{\ell-1}(I) + 1, a(G(I)) + \ell\}$$

which implies  $r_J(I) = a(G(I)) + \ell$ .

DEFINITION 4.6. Let c be an integer. We say that I has sliding reduction number c if  $r_i(I) = c + i - ht(I)$ , for  $i = ht(I), \dots, \ell(I) - 1$ .

REMARK 4.7. Set  $\ell = \ell(I)$  and s = ht(I). If  $\ell = s$ , i.e. I is an equimultiple ideal, then I can have any sliding reduction number. If  $\ell > s$ , i.e. ad(I) > 0, then I has sliding reduction number c if and only if

$$r(I_{\wp}) = c$$
 for some prime  $\wp \supseteq I$  with  $ht(\wp) = s$  and  
 $r(I_{\wp}) \leq c + ht(\wp) - s$  for every prime  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp) \leq \ell - 1$ .

Recall that the  $r_i(I)$ 's are strictly increasing. Having sliding reduction number means that the first  $\ell - 1$  increase by exactly one, the least possible. For example, I has sliding reduction number 0 if  $I_{\wp}$  is a complete intersection for every prime ideal  $\wp \supseteq I$  with  $ht(\wp) = \ell(I) - 1$ .

The following result shows that if I has sliding reduction number, then the *a*-invariant of G(I) can be computed by means of the (global) reduction numbers.

THEOREM 4.8. Assume that G(I) is a Cohen–Macaulay ring and I has sliding reduction number c. Then for any minimal reduction J of I

$$a(G(I)) = \max\{c - \operatorname{ht}(I), r_J(I) - \ell(I)\}.$$

*Proof.* Set S = G(I), s = ht(I), and  $\ell = \ell(I)$ . We have  $r_i(I) = c + i - s$ ,  $i = s, \ldots, \ell - 1$ . If  $r_J(I) - \ell > c - s$ , then  $r_J(I) > r_{\ell-1}(I) + 1$ , which is equivalent to having  $r(I_{\wp}) - ht(\wp) < r_J(I) - \ell$  for every prime  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp) < \ell$ . Hence  $a(S) = r_J(I) - \ell$  by Corollary 4.5. If  $r_J(I) - \ell \leq c - s$ , then  $r_J(I) \leq r_{\ell-1}(I) + 1$ . By Theorem 4.4 we have  $r_{\ell-1}(I) + 1 = r_{\ell}(I) \ge a(S) + \ell$ , hence

$$c - s = r_{\ell-1}(I) - \ell + 1 \ge a(S).$$

On the other hand, by [T3, Lemma 2.4] we know that

$$a(S) \ge \max\{r((S_+)_P) | P \in \operatorname{Ass}(S/S_+), \operatorname{ht}(P) = s\} - s.$$

Let  $\mathfrak{p}$  be the contraction of such a prime P in  $S_0$ . Then  $P = \mathfrak{p} \oplus S_+$  because  $P \supseteq S_+$ . Since  $\operatorname{ht}(P) = s = \operatorname{ht}(S_+)$ , we have  $\operatorname{ht}(\mathfrak{p}) = 0$ . It is easily seen that  $\ell((S_+)_P) = \ell((S_{\mathfrak{p}})_+) = s$  and  $r((S_+)_P) = r((S_{\mathfrak{p}})_+)$ . Therefore

$$\max\{r((S_+)_P)|P \in \operatorname{Ass}(S/S_+), \operatorname{ht}(P) = s\} \\ = \max\{r((S_{\mathfrak{p}})_+)|\mathfrak{p} \in \operatorname{Spec}(S_0) \text{ with } \ell((S_{\mathfrak{p}})_+) = s, \operatorname{ht}(\mathfrak{p}) = 0\} = r_s(S).$$

Therefore  $a(S) \ge r_s(S) - s$ . By Lemma 4.3,  $r_s(S) = r_s(I)$  so that  $a(S) \ge r_s(I) - s = c - s$ . Hence a(S) = c - s.  $\Box$ 

**REMARK** 4.9. Suppose that I has sliding reduction number c. The formula for a(G(I)) in Theorem 4.7 can be reformulated as follows:

$$a(G(I)) = \min\{n \ge c + \operatorname{ad}(I) | JI^n = I^{n+1}\} - \ell(I).$$

This formula has been proved for ad(I) = 1 [T3, Corollary 3.2] (cf. [GH, Proposition 2.4] for the case I is generically a complete intersection) and for ad(I) = 2 and  $I_{\wp}$  is a complete intersection for every prime  $\wp \supseteq I$  with  $ht(\wp) = ht(I) + 1$  [GN1, Theorem 1.3], [T3, Corollary 4.2].

Theorem 4.7 has the following interesting consequence on the invariance of the reduction numbers of minimal reductions.

COROLLARY 4.10. Assume that G(I) is a Cohen–Macaulay ring. If I has sliding reduction number less than or equal to r(I) - ad(I), then

 $r_J(I) = a(G(I)) + \ell(I)$ 

for any minimal reduction J of I.

*Proof.* If  $c \leq r(I) - \operatorname{ad}(I) = r(I) - \ell(I) + \operatorname{ht}(I)$  then  $\max\{c - \operatorname{ht}(I), r_J(I) - \ell(I)\} = r_J(I) - \ell(I)$ , giving the desired equality.  $\Box$ 

When G(I) is CM we can bound the Castelnuovo–Mumford regularity index of G(I) in terms of the local reduction numbers.

**PROPOSITION 4.11.** Assume that G(I) is a Cohen–Macaulay ring. Set  $\ell = \ell(I)$  and let J be an arbitrary minimal reduction of I. Then

$$\operatorname{reg}(G(I)) \leq \max\{r_{\ell-1}(I), r_J(I)\} \leq r_{\ell}(I).$$

*Proof.* Set S = G(I) and let Z be the minimal reduction of  $S_+$  generated by the initial forms of the elements of J in S. By Proposition 2.10 and Corollary 3.6(ii) we have

$$\operatorname{reg}(S) \leq \max\{a(Z) - 1, r_Z(S_+)\} \leq \max\{r_{\ell-1}(S), r_Z(S_+)\}.$$

Using Lemma 4.1 and Lemma 4.3 this can be interpreted as  $reg(S) \leq \max\{r_{\ell-1}(I), r_J(I)\}$ , which together with Theorem 4.4(i) implies the conclusion.

#### 5. Cohen-Macaulay and Gorenstein properties of Rees algebras

Throughout this section let (R, m) be a local Noetherian ring with infinite residue field. We will derive criteria for the Cohen-Macaulay and Gorenstein properties of the Rees algebra  $R[It] := \bigoplus_{n \ge 0} I^n t^n$  of an ideal I in R by means of the Cohen-Macaulay and Gorenstein properties of the associated graded ring G(I) and the reduction numbers of I.

THEOREM 5.1. Let (R, m) be a CM ring and I an ideal of R with  $ht(I) \ge 1$ . Then R[It] is CM if and only if the following conditions are satisfied:

(i) G(I) is Cohen–Macaulay.

(ii)  $r(I_{\wp}) \leq ht(\wp) - 1$  for every prime  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp)$ .

*Proof.* ( $\Rightarrow$ ) Set S = G(I). By [Hu1, Proposition 1.1], [TI, Theorem 7.1] we have (i) and a(S) < 0. Note that  $R_{\wp}[I_{\wp}t]$  is CM for every prime ideal  $\wp \supseteq I$ . Using induction on  $d = \dim(R)$  we may assume that

$$r(I_{\wp}) \leq \operatorname{ht}(\wp) - 1$$
 if  $\ell(I_{\wp}) = \operatorname{ht}(\wp) \leq d - 1$ .

To prove (ii) we need only to consider the case  $\ell(I_{\wp}) = \operatorname{ht}(\wp) = d$ . Then  $\wp = m$  and  $I_{\wp} = I$ . By the inductive assumption,  $r_{d-1}(I) \leq d-2$  (including the base case d = s). Choose  $J \subseteq I$  a minimal reduction such that  $r_J(I) = r(I)$ . Using Theorem 4.4 we get  $\max\{r_{\ell-1}(I)+1, r(I)\} = \max\{r_{\ell-1}(I)+1, a(S)+\ell\}$ . Since  $\ell = d$  we have

$$r(I) \leq \max\{r_{d-1}(I) + 1, a(S) + d\}.$$

This is at most d-1 since a(S) < 0 and  $r_{d-1}(I) \leq d-2$ .

( $\Leftarrow$ ) Set  $\ell = \ell(I)$ . By (ii) we have  $r_{\ell}(I) \leq \ell - 1$  and by Theorem 4.4,  $a(S) + \ell \leq r_{\ell}(I)$ . Thus,  $a(S) \leq r_{\ell}(I) - \ell < 0$  which together with (i) implies that R[It] is Cohen–Macaulay [TI, Theorem 7.1].

COROLLARY 5.2. Let (R, m) be a Cohen-Macaulay ring and I an ideal of R with  $ht(I) \ge 1$ . Assume that  $I_{\wp}$  is a complete intersection for every prime  $\wp \supset I$  with  $ht(\wp) = \ell(I)$ . Then R[It] is Cohen-Macaulay if and only if G(I) is Cohen-Macaulay.

*Proof.* Let  $\wp \supseteq I$  be a prime ideal of R with  $ht(\wp) = \ell(I_{\wp}) \leq \ell(I)$ . By the assumption,  $I_{\wp}$  is a complete intersection, hence  $\ell(I_{\wp}) = ht(I)$  so that  $ht(\wp) = ht(I)$ . Since  $r(I_{\wp}) = 0$ , condition (ii) of Theorem 5.1 is satisfied, and the conclusion follows.

Recall that  $P^{(n)} = \{r | \text{ there exists } s \notin P \text{ such that } sr \in P^n \}.$ 

COROLLARY 5.3. Let R be a Cohen-Macaulay ring and P a prime ideal of R with  $ht(P) \ge 1$  such that  $R_P$  is regular. Assume that  $P^{(n)} = P^n$  for  $n \ge 1$ . Then R[Pt] is Cohen-Macaulay if and only if G(P) is Cohen-Macaulay.

*Proof.* It is known [CN] that the assumption  $P^{(n)} = P^n$  for  $n \ge 1$  implies  $\ell(P_{\wp}) < ht(\wp)$  for every prime ideal  $\wp \supseteq P$  of R. Hence we need to check condition (ii) of Theorem 5.1 only for  $\wp = P$ . Since  $R_P$  is regular,  $PR_P$  is a complete intersection, hence the conclusion follows.

Corollary 5.3 is proved for R Gorenstein in [Va3, Theorem 5.7.7].

Recently Lipman [L] has proved the following result (We refer to [LT] for a discussion of and definition of pseudo-rational rings. However note that by [LT], regular rings are pseudo-rational):

THEOREM 5.4. Let (R, m) be a pseudo-rational local ring. Let I be any ideal of R. Then G(I) is CM if and only if R[It] is CM.

Combining this result with Theorem 5.1 gives us

**PROPOSITION 5.5.** Let (R, m) be a CM local ring and I an ideal of R. Assume that  $R_P$  is pseudo-rational whenever  $\ell(I_P) = \dim R_P$ . Then G(I) is CM if and only if R[It] is CM.

*Proof.* [Hu, Proposition 1.1] shows that G(I) is CM if R[It] is CM. Assume that G(I) is CM. By Theorem 5.1 we need only show that  $r(I_P) \leq ht(P) - 1$  whenever  $\ell(I_P) = ht(P)$ . By our assumption,  $R_P$  is regular at such primes P, so by Theorem 5.4,  $R_P[I_Pt]$  is CM, therefore Theorem 5.1 applied to  $R_P[I_Pt]$  gives the desired conclusion.

In fact, Lipman uses his methods to prove (5.5) also, but we noticed it independently and the proof points out the efficacy of (5.1).

A slightly different version of Theorem 5.1 is the following criterion for the Cohen-Macaulay property of R[It] which involves the reduction number of I. This criterion generalizes results in the cases ad(I) = 0 [GS], [GHO]; ad(I) = 1 [HuHu1], [GH], [Vi], [T3]; ad(I) = 2 [GN1], [T3].

THEOREM 5.6. Let R be a Cohen–Macaulay ring and I an ideal of R with  $ht(I) \ge 1$ . Then R[It] is Cohen–Macaulay if and only if the following conditions are satisfied:

(i) G(I) is Cohen–Macaulay.

(ii)  $r(I_{\wp}) \leq ht(\wp) - 1$  for every prime  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp) < \ell(I)$ .

(iii)  $r_J(I) \leq \ell(I) - 1$  for some (equivalently, every) minimal reduction J of I.

*Proof.* ( $\Rightarrow$ ) Set  $\ell = \ell(I)$ . By Theorem 5.1 we already have (i), (ii), and  $r_{\ell}(I) \leq \ell - 1$ . By Theorem 4.4 (i) the latter relation implies  $r_J(I) \leq \ell - 1$ , hence (iii) holds.

(⇐) By Theorem 5.1 we need only to show that  $r(I_{\wp}) \leq \ell - 1$  for every prime  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp) = \ell$ . But this follows from (iii) because  $J_{\wp}$  is a minimal reduction of  $I_{\wp}$  so that  $r(I_{\wp}) \leq r_J(I)$ .  $\Box$ 

COROLLARY 5.7. Let R be a Cohen–Macaulay ring and I an ideal of R with  $ht(I) \ge 1$ . Assume that  $I_{\wp}$  is a complete intersection for every prime  $\wp \supseteq I$  with  $ht(\wp) < \ell(I)$ . Then R[It] is Cohen–Macaulay if and only if the following conditions are satisfied:

(i) G(I) is Cohen–Macaulay.

(ii)  $r_J(I) \leq \ell(I) - 1$  for some (equivalently, every) minimal reduction J of I.

*Proof.* Condition (ii) of Theorem 5.6 is satisfied by the assumption on I, since  $r(I_{\wp}) = 0$  for all  $\wp \supseteq I$  such that  $ht(\wp) < \ell$ .

THEOREM 5.8. Let R be a Cohen–Macaulay ring and I an ideal of R with  $ht(I) \ge 2$ . Then R[It] is Gorenstein if and only if the following conditions are satisfied:

(i) G(I) is Gorenstein.

(ii)  $r(I_{\wp}) = ht(\wp) - 2$  for every prime ideal  $\wp \supseteq I$  with  $ht(\wp/I) = 0$ .

(iii)  $r(I_{\wp}) \leq ht(\wp) - 2$  for every prime ideal  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp) \leq \ell(I)$ .

*Proof.* ( $\Rightarrow$ ) Set S = G(I). By [I, Theorem 3.1] and [TI, Theorem 7.1] we have (i) and a(S) = -2. Note that  $R[I_{\wp}t]$  is Gorenstein for every prime ideal  $\wp \supseteq I$ . To see (ii), let  $\wp$  be a minimal prime of I. After localizing we may assume that I is *m*-primary, so  $\ell = d = \dim S$ . Since  $r_{d-1}(I) = -1$  we can apply Theorem 4.4(i) to get  $r_d(I) = \max\{-1 + 1, r(I)\}$  so  $r(I) = r_d(I)$ . Then by Theorem 4.4(ii)

$$r(I) = \max\{r_{d-1}(I) + 1, a(S) + d\} = d - 2.$$

This gives (ii).

We now prove (iii) by induction on  $d = \dim S$ . If d = s then (ii) gives us the result. Assume now that the inductive hypothesis holds for dimension smaller than

*d*. If  $\ell(I) < d$  then (iii) holds by induction. Thus we may assume that  $\ell(I) = d$ . For  $\wp \subsetneq m$ , the induction hypothesis shows that (iii) holds and also that  $r_i(I) = i - 2$  for  $s \leqslant i < \ell$ . Hence I has sliding reduction number s - 2. By Theorem 4.8

$$-2 = a(S) = \max\{-2, r(I) - d\}.$$

From this it follows that  $r(I) \leq d - 2$ , which proves (iii).

( $\Leftarrow$ ) By (ii) and (iii) we have  $r_i(I) = i - 2$  for  $i = s, \ldots, \ell$ . Hence I has sliding reduction number s - 2 and  $r(I) \leq r_\ell(I) = \ell - 2$  by Theorem 4.4(i). Using Theorem 4.8 we get

$$a(S) = \max\{-2, r(I) - \ell\} = -2$$

which together with (i) implies by [I, Theorem 3.1] and [TI, Theorem 7.1] that R[It] is Gorenstein.

REMARK 5.9. Suppose R[It] is Gorenstein. If I is generically a complete intersection, that is,  $I_{\wp}$  is a complete intersection for every prime ideal  $\wp \supseteq I$  with  $\dim R/\wp = \dim R/I$ , then  $r(I_{\wp}) = 0$  and  $\operatorname{ht}(\wp/I) = 0$  for such a prime  $\wp$ . Hence condition (ii) of Theorem 5.8 implies that  $\operatorname{ht}(\wp) = 2$  so that  $\operatorname{ht}(I) = 2$  (cf. [I, Corollary 4.5]).

COROLLARY 5.10. Let R be a Cohen–Macaulay ring and I an ideal of R with  $ht(I) \ge 2$ . Assume that  $I_{\wp}$  is a complete intersection for every prime  $\wp \supseteq I$  with  $ht(\wp) = \ell(I)$ . Then R[It] is Gorenstein if and only if ht(I) = 2, R/I is equidimensional, and G(I) is Gorenstein.

*Proof.* ( $\Rightarrow$ ) If P is minimal over I then ht(P)  $\leq \ell(I)$ . Hence  $I_P$  is a complete intersection and Theorem 5.8(ii) shows that ht(P) = 2. Thus R/I is equidimensional and ht(I) = 2.

( $\Leftarrow$ ) The only primes for which  $\ell(I_P) = \operatorname{ht}(P) \leq \ell(I)$  are the minimal primes of *I*. Since then  $I_P$  is a complete intersection, conditions (ii) and (iii) are satisfied.  $\Box$ 

COROLLARY 5.11. Let R be a Cohen-Macaulay ring and P a prime ideal of R with  $ht(P) \ge 2$  such that  $R_P$  is regular. Assume that  $P^{(n)} = P^n$  for  $n \ge 1$ . Then R[Pt] is Gorenstein if and only if ht(P) = 2 and G(P) is Gorenstein.

*Proof.* The hypothesis that  $P^n = P^{(n)}$  for  $n \ge 1$  implies that  $\ell(I_{\wp}) < \operatorname{ht}(\wp)$  for every  $\wp \supseteq P$ . Since  $PR_P$  is a complete intersection of dimension ht P, we have R[Pt] Gorenstein if and only if  $\operatorname{ht}(P) = 2$ .

The Gorenstein property of R[It] can be also expressed by means of the reduction number of I. The following criterion generalizes results in the cases ad(I) = 0 [I]; ad(I) = 1 [GH], [T3]; and ad(I) = 2 [GN2], [T3]. See also [HHR].

THEOREM 5.12. Let R be a Cohen–Macaulay ring and I an ideal of R with  $ht(I) \ge 2$ . Then R[It] is Gorenstein if and only if the following conditions are satisfied:

- (i) G(I) is Gorenstein.
- (ii)  $r(I_{\wp}) = ht(\wp) 2$  for every prime ideal  $\wp \supseteq I$  with  $ht(\wp/I) = 0$ .
- (iii)  $r(I_{\wp}) \leq ht(\wp) 2$  for every prime ideal  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp) < \ell(I)$ .
- (iv)  $r_J(I) \leq \ell(I) 2$  for some (equivalently, every) minimal reduction J of I.

*Proof.* ( $\Rightarrow$ ) Set  $\ell = \ell(I)$ . By Theorem 5.8 we already have (i), (ii), (iii) and  $r_{\ell}(I) = \ell - 2$ . By Theorem 4.4(i),  $r_J(I) \leq r_{\ell}(I) \leq \ell - 2$  for any reduction  $J \subseteq I$ , giving (iv).

(⇐) By Theorem 5.8 we need only to show that  $r(I_{\wp}) \leq \ell - 2$  for every prime  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp) = \ell$ . But this follows from (iv) because if  $J \subseteq I$  is a minimal reduction then  $J_{\wp}$  is a minimal reduction of  $I_{\wp}$  so that  $r(I_{\wp}) \leq r_J(I)$ .  $\Box$ 

REMARK 5.13. We may replace condition (iii) of Theorem 5.12 by the condition that I has sliding reduction number ht(I) - 2.

COROLLARY 5.14. Let R be a Cohen–Macaulay ring and I an ideal of R with  $ht(I) \ge 2$ . Assume that  $I_{\wp}$  is a complete intersection for every prime  $\wp \supseteq I$  with  $ht(\wp/I) = 0$  and for all  $\wp \supseteq I$  such that  $ht(\wp) < \ell(I)$ . Then R[It] is Gorenstein if and only if ht(I) = 2, R/I is equidimensional and the following conditions are satisfied:

(i) G(I) is Gorenstein.

(ii)  $r_J(I) \leq \ell - 2$  for some (equivalently, every) minimal reduction J of I.

## 6. Castelnuovo–Mumford regularity and Serre properties of Rees algebras

Throughout this section let (R, m) be a Noetherian local ring with infinite residue field. We will study the Castelnuovo–Mumford regularity and the Serre properties of the Rees algebra R[It] of an ideal I in R.

LEMMA 6.1. Let I be a proper ideal of R with  $ht(I) \ge 1$ . Then  $\ell(R[It]_+) = \ell(I)$ and  $r_i(R[It]) = i - 1$  for  $i = 1, ..., \ell(I)$ .

*Proof.* Set T = R[It]. The first statement follows from the fact that  $T_n/mT_n \simeq I^n/mI^n$ ,  $n \ge 0$ . Recall that if  $ht(I) \ge 1$  then dim  $T = \dim R + 1$  and  $ht(T_+) = 1$  (see [Ma]). For the second statement we note that  $ht(T_+) = 1$ . If  $\wp \supseteq I$  is a prime ideal of R, then  $\ell((T_{\wp})_+) = \ell(I_{\wp}) \le ht(\wp)$ . If  $\wp \not\supseteq I$ , then  $T_{\wp} = R_{\wp}[t]$ , hence  $\ell((T_{\wp})_+) = 1$  and  $r((T_{\wp})_+) = 0$ . Therefore, by the definition of  $r_i(T)$  we get

$$\begin{split} r_i(T) &= \max\{r((T_{\wp})_+) - \operatorname{ht}(\wp) | \ell((T_{\wp})_+) = \dim T_{\wp} = \operatorname{ht}(\wp) + 1 \leqslant i\} + i - 1 \\ &= \max\{r((T_{\wp})_+) | \operatorname{ht}(\wp) = 0\} + i - 1 = i - 1. \end{split}$$

**PROPOSITION 6.2.** Let I be a proper ideal of R with  $ht(I) \ge 1$ . Assume that R[It] is a Cohen–Macaulay ring. Then

 $\operatorname{reg}(R[It]) \leq \ell(I) - 1.$ 

*Proof.* We have by Proposition 2.10 and Corollary 3.6, that  $\operatorname{reg}(R[It]) \leq r_{\ell}(R[It])$  and by Lemma 6.1,  $r_{\ell}(R[It]) = \ell(I) - 1$ .

Recall that the relation type N(I) of R[It] is the maximal degree of the elements of a homogeneous minimal basis of the defining ideal of R[It] represented as a quotient ring of a polynomial ring over R.

COROLLARY 6.3. Let I be an ideal of R with  $ht(I) \ge 1$ . Assume that R[It] is a Cohen–Macaulay ring. Then

 $N(I) \leq \ell(I).$ 

Proof. See Remark 2.11.

We show now that when R is assumed to be CM as well (in which case G(I) will be CM) then N(I) can be bounded in terms of the local reduction numbers of I.

LEMMA 6.4. Let (R, m) be a CM ring and let I be an ideal of height s and analytic spread  $\ell$ . Let  $J = (a_1, \ldots, a_\ell)$  be a minimal reduction of I. If G(I) is CM and  $a_1^*, \ldots, a_\ell^*$  is an  $[0, \ldots, 0, n_{s+1}, \ldots, n_\ell]$ -regular sequence where  $n_{s+1} \leq \cdots \leq n_\ell$ then for all  $1 \leq k \leq \ell$  and all  $n \geq n_k + 2$  we have

 $(a_1,\ldots,a_k)I^{n-2}\cap I^n\subseteq (a_1,\ldots,a_k)I^{n-1}.$ 

*Proof.* Let  $J_k = (a_1, \ldots, a_k)$ . If  $k \leq s$  then  $a_1, \ldots, a_k$  and  $a_1^*, \ldots, a_k^*$  are regular sequences so  $J_k \cap I^n = J_k I^{n-1}$  for all n. So assume that k > s and that the claim is true for k - 1.

Let  $n \ge n_k + 2$  and let  $x \in J_k I^{n-2} \cap I^n$ . Say  $x = a_1 i_1 + \cdots + a_k i_k$  where  $i_j \in I^{n-2}$ . If  $i_k \in I^{n-1}$  then we are done by induction. If not then

 $i_k^* \in [(a_1^*, \dots, a_{k-1}^*) : a_k^*]_{n-2} = [(a_1^*, \dots, a_{k-1}^*)]_{n-2}$ 

since  $n-2 \ge n_k$ . Thus  $i_k \in J_{k-1}I^{n-3} + I^{n-1}$ . Hence  $x = a_1i'_1 + \cdots + a_{k-1}i'_{k-1} + a_ky \in J_{k-1}I^{n-2} + a_kI^{n-1}$ . Then  $a_1i'_1 + \cdots + a_{k-1}i'_{k-1} \in J_kI^{n-2} \cap I^n \subseteq J_{k-1}I^{n-1}$ since  $n \ge n_k \ge n_{k-1}$ . Hence  $x \in J_kI^{n-1}$ .

**PROPOSITION 6.5.** Let (R, m) be a CM ring and I an ideal of height s and analytic spread  $\ell$ . Let  $J = (a_1, \ldots, a_\ell)$  be a minimal reduction of I. Suppose that R[It] is CM. If  $a_1^*, \ldots, a_\ell^*$  is  $[0, \ldots, 0, n_{s+1}, \ldots, n_\ell]$ -regular in G(I), where  $n_{s+1} \leq \cdots \leq n_\ell$ , then  $a_1t, \ldots, a_\ell t$  is  $[0, 1, \ldots, 1, n_{s+1}, \ldots, n_\ell]$ -regular in R[It](there are s - 1 1's).

*Proof.* Let  $J_k = (a_1, \ldots, a_k)$ . Since  $a_1$  is *R*-regular we have that  $a_1t$  is *R*[*It*]-regular. Now suppose that  $1 < k \leq s, n \geq 1$  and  $xt^n \in [(a_1t, \ldots, a_{k-1}t) :$ 

 $a_k t]_n$ . Then  $x \in I^n \cap J_k$ :  $a_k = I^n \cap J_k = J_k I^{n-1}$  so  $xt^n \in J_k I^{n-1}t^n = (J_k t)I^{n-1}t^{n-1} = (J_k t)_n$ .

Assume now that k > s. Let  $n \ge n_k$  and suppose  $(xt^n)(a_kt) \in (a_1t, \ldots, a_{k-1}t)_{n+1}$ . Then  $xa_k \in J_{k-1}I^n$ . Interpreting this in G(I) gives  $x^* \in [(a_1^*, \ldots, a_{k-1}^*)]$ :  $a_k^*]_n = (a_1^*, \ldots, a_{k-1}^*)_n$ . Hence  $x \in J_{k-1}I^{n-1} + I^{n+1}$ , say  $x = a_1i_1 + \cdots + a_{k-1}i_{k-1} + y$  where  $i_j \in I^{n-1}$  and  $y \in I^{n+1}$ . Then  $a_ky \in J_{k-1}I^n \cap I^{n+2} \subseteq J_{k-1}I^{n+1}$  by Lemma 6.4 since  $n \ge n_{k-1}$ . By the same argument as for x we obtain  $y \in J_{k-1}I^n + I^{n+2}$ . Continuing this way we get  $x \in \cap_{m \ge n}(J_{k-1}I^{n-1} + I^m) = J_{k-1}I^{n-1}$ , so  $xt^n \in (a_1t, \ldots, a_{k-1}t)_n$ .

COROLLARY 6.6. With R, I, J and  $1 \leq n_{s+1} \leq \cdots \leq n_{\ell}$  as in Proposition 6.5 we have  $\operatorname{reg}(R[It]) \leq \max\{n_{s+1} - 1, \ldots, n_{\ell} - 1, r_J(I)\}$ . In particular  $\operatorname{reg}(R[It]) \leq \max\{r_{\ell-1}(I), r(I)\}$ .

*Proof.* By Proposition 2.10 we have  $\operatorname{reg}(R[It]) \leq \max\{a(Jt) - 1, r_{Jt}(It)\}$ . But  $r_{Jt}(It) = r_J(I)$  and  $a(Jt) \leq \max\{1, n_{s+1}, \ldots, n_\ell\}$  by Proposition 6.5. This gives the desired bound.

Now, since G(I) is CM we have  $n_j \leq r_{j-1}(I) + 1$  for  $1 \leq j \leq \ell$  by Theorem 3.2. Hence  $\operatorname{reg}(R[It]) \leq \max\{r_{\ell-1}(I), r(I)\}$  by applying Proposition 6.5 with J a minimal reduction such that  $r(I) = r_J(I)$ .

COROLLARY 6.7. Let R and R[It] be CM rings. Then  $N(I) \leq \max\{r_{\ell-1}(I) + 1, r(I) + 1\} \leq r_{\ell}(I) + 1$ .

*Proof.* This follows from Remark 2.11 and Corollary 3.6 (iii).

The bound for N(I) given in Corollary 6.7 is proved in the case ad(I) = 0 or ad(I) = 1 and I is generically a complete intersection [HuHu1], [GH].

Now we will use Theorem 5.1 to compare the Serre condition  $(S_k)$  in the associated graded ring and the Rees algebra. Recall that the Serre condition  $(S_k)$  on a Noetherian ring T is defined by the condition that

 $depth(T_P) \ge min\{ht(P), k\}$ 

holds for every prime ideal P of T. We are here inspired by a result of Noh and Vasconcelos [NV, Theorem 2.2].

THEOREM 6.8. Let R be a Noetherian ring satisfying  $(S_{k+1})$  and let I be an ideal of R with  $ht(I) \ge 1$ . Then R[It] satisfies  $(S_{k+1})$  if and only if the following conditions are satisfied:

(i) G(I) satisfies  $(S_k)$ .

(ii)  $r(I_{\wp}) \leq ht(\wp) - 1$  for every prime  $\wp \supseteq I$  with  $\ell(I_{\wp}) = ht(\wp) \leq k$ .

*Proof.* ( $\Rightarrow$ ) Set S = G(I). As in the proof of [NV, Theorem 2.2], to show (i) we only need to prove that  $S_P$  is Cohen-Macaulay for every prime  $P \supseteq S_+$ of S with  $ht(P) \leq k$ . Note that S = R[It]/IR[It] and let Q be the inverse image of P in R[It]. Then  $Q \supseteq R[It]_+$  and  $ht(Q) \leq k + 1$ . Hence  $R[It]_Q$  is a

Cohen-Macaulay ring by the condition  $(S_{k+1})$  of R[It]. Let  $\varphi = Q \cap R$ . Then  $R[It]_Q$  is the localization of  $R_{\varphi}[I_{\varphi}t]$  at its (unique) maximal graded ideal. Hence  $R_{\varphi}[I_{\varphi}t]$  is Cohen-Macaulay [HR, MR]. Note that  $ht(\varphi) \leq ht(Q) \leq k+1$ . Then  $R_{\varphi}$  is Cohen-Macaulay since R satisfies  $(S_k)$ . Hence  $G(I_{\varphi})$  is Cohen-Macaulay by [Hu, Proposition 1.1]. Since  $S_P$  is a localization of  $G(I_{\varphi})$ ,  $S_P$  is Cohen-Macaulay too.

To prove (ii) let  $\wp \supseteq I$  be a prime ideal of R with  $\ell(I_{\wp}) = \operatorname{ht}(\wp) \leq k$  and  $Q = (\wp, It)$ . Then  $\operatorname{ht}(Q) \leq \operatorname{ht}(\wp) + 1 \leq k + 1$ . Similarly as above we can show that  $R_{\wp}[I_{\wp}t]$  is Cohen-Macaulay. Since  $R_{\wp}$  is Cohen-Macaulay, we can apply Theorem 5.1 to deduce that  $r(I_{\wp}) \leq \operatorname{ht}(\wp) - 1$ .

(⇐) Let Q be an arbitrary prime ideal of R[It] and  $\wp = Q \cap R$ . Then  $ht(\wp) \leq ht(Q)$ . Note that  $R[It]_Q$  is a localization of  $R_{\wp}[I_{\wp}t]$ . If  $\wp \not\supseteq I$ , then  $R_{\wp}[I_{\wp}t] = R_{\wp}[t]$ . Since  $R_{\wp}$  has  $(S_{k+1})$ , so has  $R_{\wp}[t]$ . Hence

depth  $R[It]_Q \ge \min\{\operatorname{ht}(Q), k+1\}.$ 

Now let  $\wp \supseteq I$ . If  $ht(\wp) \leq k$ , then  $R_{\wp}$  is a Cohen–Macaulay ring. Since  $\dim G(I_{\wp}) = ht(\wp) \leq k$ , by (i) we have that  $G(I_{\wp})$  is a Cohen–Macaulay ring. So we can apply Theorem 5.1 to deduce from (ii) that  $R_{\wp}[I_{\wp}t]$  and hence  $R[It]_Q$  is a Cohen–Macaulay ring. If  $ht(\wp) \geq k + 1$ , we proceed as in the proof for ( $\Leftarrow$ ) of [NV, Theorem 2.2] to show that depth  $R[It]_Q \geq \min\{ht(Q), k+1\}$ .  $\Box$ 

#### REMARK 6.9.

- (1) To prove  $(\Rightarrow)$  we only need condition  $(S_k)$  on R.
- (2) It is assumed in [NV, Theorem 2.2] that ht(I) ≥ k + 1. In this case condition (ii) of Theorem 6.8 is automatically satisfied.

THEOREM 6.10. Let (R, m) be a regular local ring. Let I be an ideal of R of analytic spread  $\ell$ . If G(I) satisfies  $(S_{\ell})$  then R[It] satisfies  $(S_{\ell+1})$ .

*Proof.* By Theorem 6.8 we need to show that  $r(I_P) \leq ht(P) - 1$  for every prime  $P \supseteq I$  such that  $\ell(I_P) = ht(P) \leq \ell$ . This follows from Theorems 5.1 and 5.4.

## 7. Briançon–Skoda theorem with coefficients for regular rings containing a field

The purpose of this section is to give an improved version of the Briançon–Skoda theorem in [AH, Sect. 3]. The point of the improvement is to obtain more accurate information about coefficients for ideals of higher analytic deviation. Since the proof uses tight closure methods we will obtain the results only for regular local rings containing a field. The case of mixed characteristic is still open. The results from this section will be applied in Section 8 in order to bound reduction numbers

for ideals of small analytic deviation when G(I) satisfies Serre's condition  $(S_{\ell})$ , where  $\ell = \ell(I)$ .

First we shall establish that given any ideal I in any local ring (R, m) with infinite residue field, we may pick generators of I such that the first i are a reduction of  $I_P$  for any prime  $P \supseteq I$  of height i. The idea behind this comes from Lemma 2.3 of [AN] where good generating sets of I are chosen subject to the condition that  $\mu(I_P) \leq ht P$  up to a certain height. Since we are choosing reductions we are using the "hypothesis" that  $\ell(I_P) \leq ht P$  for primes P up to a certain height. But by the nature of analytic spread this is always satisfied.

DEFINITION 7.1. Let (R, m) be any local ring and let  $I \subseteq R$  be an ideal and let J be a minimal reduction of I. Then the generating set  $a_1, \ldots, a_n$  of J is a *basic* generating set for J if for all primes  $P \supseteq I$  such that  $i = \operatorname{ht} P < \ell(I)$  we have that  $(a_1, \ldots, a_i)_P$  is a reduction of  $I_P$ .

We use the term "basic" to suggest the parallels with basic element theory.

The following lemma allows us to make the Briançon–Skoda Theorem statements.  $\ensuremath{^\circ}$ 

LEMMA 7.2. Let (R, m) be a local ring with infinite residue field. Let  $I \subseteq R$  be any ideal and let  $\ell = \ell(I)$ . For every minimal reduction J of I, there exists a minimal generating set  $a_1, \ldots, a_\ell$  such that

- (1) if P is a prime ideal containing I and ht  $P = i < \ell$  then  $(a_1, \ldots, a_i)_P$  is a reduction of  $I_P$ , and
- (2)  $ht((a_1, \ldots, a_i)I^n : I^{n+1} + I) \ge i + 1$  for all  $n \gg 0$ .

*Proof.* Let G = G(I). Note that  $(a_1, \ldots, a_\ell)$  is a reduction of I if and only if  $(a_1^*, \ldots, a_\ell^*)$  is a reduction of  $G_+$ . Fix a minimal reduction J of I. Choose  $a_1, \ldots, a_\ell$  a minimal generating set of J such that  $a_1^*, \ldots, a_\ell^*$  is a filter regular sequence with respect to  $G_+$  (see [T2]). Let  $P \supseteq I$  be a prime of height  $i < \ell$ . Then  $a_1^*, \ldots, a_\ell^*$  is still filter regular in  $G_P$  with respect to  $(G_P)_+$ . Since dim  $G_P =$ ht P = i, we may apply Corollary 2.2 and Lemma 2.4 to see that  $(a_1^*, \ldots, a_i^*)_P$  is a reduction of  $(G_P)_+$ . Hence  $(a_1, \ldots, a_i)_P$  is a reduction of  $I_P$ .

To see (2) we note that if ht P < i + 1 then  $(a_1, \ldots, a_i)_P$  is a reduction of  $I_P$  so P cannot contain  $\bigcup_n ((a_1, \ldots, a_i)I^n) : I^{n+1} + I$ .  $\Box$ 

REMARK 7.3. There is an alternative approach to the proof of Lemma 7.2 which follows along the lines of the proof of [AN, Lemma 2.3]. Given an ideal I and  $K \subseteq I$  we can define

$$L_i(K, I) = \sum_{x_1, \dots, x_{i-1} \in I} \bigcup_{n \ge 0} (K + (x_1, \dots, x_{i-1})) I^n : I^{n+1},$$

for all *i*. So if  $J \subseteq I$  is a minimal reduction we pick a minimal generating set  $a_1, \ldots, a_\ell$  inductively such that letting  $J_r = (a_1, \ldots, a_r)$  for  $r < \ell$  and letting  $L_i^r = L_i(J_r, I)$  we have

- (1) ht  $L_i^r \ge i + r$  for i > 1 and  $r + i < \ell$ , (2) ht $(L_1^r + I) \ge r + 1$ , and
- (3) ht  $L_1^r \ge r$ .

In order to state our Briançon–Skoda theorem we will need some terminology indicating the removal of primary components whose height is too large. In other words, we need a refinement of the notion of  $I^{\text{un}}$  given in [AH, Definition 3.1].

DEFINITION 7.4. Fix an ideal K of a ring R. Let I be an ideal and let  $t > \operatorname{ht} K$  be an integer. Let  $I^{\langle t,K \rangle} = S^{-1}IR \cap R$  where  $S = R - \bigcup \{P \in \operatorname{Ass}(R/K) | \operatorname{ht}(P) < t\}$ . Often, if the ideal K is understood we will write simply  $I^{\langle t \rangle}$ . The ideal  $I^{\langle t,K \rangle}$ is the intersection of the primary components of I whose radicals are contained in an associated prime of K of height less than t.

REMARK 7.5. If R is a regular ring of characteristic p and A is an ideal of R then  $(A^{(j,K)})^{[q]} = (A^{[q]})^{(j,K)}$ .

*Proof.* We note that for any ideals I and J,  $(I \cap J)^{[q]} = I^{[q]} \cap J^{[q]}$  and if I is P-primary then  $I^{[q]}$  is also P-primary, since the Frobenius endomorphism is flat. Let  $A = W_1 \cap \cdots \cap W_n \cap V_1 \cap \cdots \cap V_m$  be a primary decomposition where  $\sqrt{W_i}$  is contained in an associated prime of K of height less than j, and the  $V_i$  are the other components. Then  $A^{\langle j \rangle} = W_1 \cap \cdots \cap W_n$  and  $A^{[q]} = W_1^{[q]} \cap \cdots \cap W_n^{[q]} \cap V_1^{[q]} \cap \cdots \cap V_m^{[q]}$ , so  $(A^{[q]})^{\langle j \rangle} = W_1^{[q]} \cap \cdots \cap W_n^{[q]} = (A^{\langle j \rangle})^{[q]}$ .

THEOREM 7.6. Let (R, m) be a regular local ring containing a field. Let I be any ideal of R of height g and analytic spread  $\ell$ . Let  $J = (a_1, \ldots, a_\ell)$  be a minimal reduction generated by a basic generating set. For each  $i \ge g$  let  $J_i = (a_1, \ldots, a_i)$ . Then for every  $w \ge 0$ 

$$\overline{I^{\ell+w}} \subseteq J^{w+1}(J_{\ell-1}(\cdots(J_g)^{\langle g+1,I\rangle}\cdots)^{\langle \ell-1,I\rangle})^{\langle \ell,I\rangle}.$$

*Proof.* We give the proof in characteristic p > 0 and then the result follows in equicharacteristic 0 by standard methods of reduction to characteristic p. See [AH] for a thorough treatment of a similar reduction.

For  $g \leq i < \ell$  let  $A_{i+1} = \bigcup_n (J_i J_{i+1}^n : J_{i+1}^{n+1})$ . Note that by Lemma 7.2 (1) we have ht  $A_{i+1} + I \geq i + 1$ . Let  $n_i$  be the value where the union stabilizes. Then for any prime P of height i + 1 containing I we have  $r_{(J_i)P}((J_{i+1})P) \leq n_i$ . For each  $q = p^e$  let  $A_{i+1,q} = (J_i^{iq} J_{i+1}^{n_i} : J_{i+1}^{iq+n_i})$ . Note that for all  $q, \sqrt{A_i} = \sqrt{A_{i,q}}$ . Now choose  $f_g, \ldots, f_{\ell-1}$  such that  $f_i \in J_{i+1}^{n_i}$  for each  $g \leq i < \ell$ .

Let  $d \neq 0$  be an element such that  $dI^t \subseteq J^t$  for all t (e.g.  $d \in I^{r_J(I)}$  will work since J is a reduction of I). Now let x be a nonzero element of  $\overline{I^{\ell+w}}$ . Then letting  $c = x^k$  for k large enough we have  $cdx^q \in dI^{(\ell+w)q} \subseteq J^{(\ell+w)q}$ . Then by Lemma 7.8 below

$$f_g \cdots f_{\ell-1} c dx^q \in f_g \cdots f_{\ell-1} J^{(\ell+w)q} \subseteq (J^{w+1})^{[q]} f_g \cdots f_{\ell-1} J^{(\ell-1)q}$$

We wish to show that

$$f_g \cdots f_{\ell-1} J^{(\ell-1)q} \subseteq (J_{\ell-1} (J_{\ell-2} \cdots (J_g)^{\langle g+1 \rangle} \cdots)^{\langle \ell-1 \rangle})^{\langle \ell \rangle [q]}.$$

It then follows that x is in the tight closure of the desired ideal and since all ideals in a regular ring are tightly closed we are done.

We first note that  $A_{i+1,q}f_iJ_{i+1}^{iq} \subseteq J_i^{[q]}J_i^{(i-1)q}$ . In particular,  $A_{g+1,q}f_gJ_{g+1}^{gq} \subseteq J_g^{[q]}$ . Now, since ht  $A_{g+1,q} + I \ge g+1$ , the ideal  $A_{g+1,q}$  is not in any associated prime of  $(J_g^{[q]})^{(g+1)}$ . Hence  $f_gJ_{g+1}^{gq} \subset (J_g^{[q]})^{(g+1)}$ . Suppose inductively that

$$f_{i-1}\cdots f_g J_i^{(i-1)} \subseteq (J_{i-1}^{[q]}\cdots (J_g^{[q]})^{\langle g+1\rangle}\cdots)^{\langle i\rangle},\tag{*}$$

for some *i* with  $g \leq i - 1 < \ell - 1$ . Then

$$A_{i+1,q}f_{i}\cdots f_{g}J_{i+1}^{iq} \subseteq f_{i-1}\cdots f_{g}J_{i}^{[q]}J_{i}^{(i-1)q}$$
$$\subseteq J_{i}^{[q]}(J_{i-1}^{[q]}\cdots (J_{g}^{[q]})^{\langle g+1\rangle}\cdots)^{\langle i\rangle}$$

(the second inclusion comes from multiplying (\*) by  $J_i^{[q]}$ ). Now, since ht $(A_{i+1,q} + I) \ge i+1$  we get that

$$f_i \cdots f_g J_{i+1}^{iq} \subseteq (J_i^{[q]} (\cdots)^{\langle i \rangle})^{\langle i+1 \rangle}.$$

In particular  $f_{\ell-1} \cdots f_g J^{(\ell-1)g} \subseteq (J_{\ell}^{[q]} \cdots)^{\langle \ell \rangle}$  and Remark 7.5 gives the desired result.

REMARK 7.7. This result is stronger than [AH, Theorem 3.3], especially in the case where the ideal I has minimal components of different heights.

LEMMA 7.8. [AH, Lemma 3.4]. Let R be any commutative ring and  $A = (u_1, \ldots, u_n) \subseteq R$ . Then for all  $w, k \ge 0$  we have

$$A^{(n+w)k} \subseteq (u_1^k, \dots, u_n^k) A^{(n-1)k}.$$

#### 8. Reduction numbers of ideals in regular rings

We can now apply the results of Sections 6 and 7 to ideals of small analytic deviation in regular rings containing a field to get bounds on reduction numbers. The main results of this section are Theorems 8.4 and 8.5, in which we prove that, under certain conditions on I,  $r(I) < \ell(I)$  if G(I) satisfies Serre's condition  $(S_{\ell(I)})$ . We will use repeatedly the result of [VV] that if  $g = \operatorname{ht} I$  and  $a_1, \ldots, a_g \in I - I^2$  are a regular sequence then  $a_1^*, \ldots, a_g^*$  is G(I)-regular if and only if  $(a_1, \ldots, a_g) \cap I^n =$  $(a_1, \ldots, a_g)I^{n-1}$  for all n. For the rest of this section, if I is an ideal and  $J = (a_1, \ldots, a_\ell)$  is a minimal reduction generated by a basic generating set then we will let  $J_i = (a_1, \ldots, a_i)$  and  $K_i = (J_i \cdots (J_g)^{\langle g+1, I \rangle} \cdots)^{\langle i+1, I \rangle}$  for  $g \leq i \leq \ell - 1$ . For any ideal  $K \subseteq R$  we use  $\operatorname{Min}^0(K)$  to denote the set of minimal primes of K which have maximal dimension.

We will need notation to handle the image of an element  $u \in R$  in G(I) when we do not know precisely the degree of u.

DEFINITION 8.1. Let  $I \subseteq R$  and  $u \in R$ . If  $u \in I^j$  then we denote by  $u^{*(j)}$  the element  $u + I^{j+1}$  considered to be in the *jth* graded piece of G(I). Note that when  $u \in I^j - I^{j+1}$  we have  $u^{*(j)} = u^*$  and when  $u \in I^{j+1}$  then  $u^{*(j)} = 0$ .

**PROPOSITION 8.2.** Let (R, m) be a CM ring with infinite residue field and let  $I \subseteq R$  with  $\ell = \ell(I)$  and  $g = \operatorname{ht} I > 0$ . If I has sliding reduction number zero, G(I) is CM, and  $I^{\ell} \subseteq JI^{\ell-g}$  for some minimal reduction J then R[It] is CM.

*Proof.* Since I has sliding reduction number zero we need only show that  $r(I) < \ell$ , in order to apply Theorem 5.6. Let  $a_1, \ldots, a_\ell$  be a basic generating set of the reduction J and let  $J_i = (a_1, \ldots, a_i)$  for  $1 \le i \le \ell$ . Because I has sliding reduction number zero, we know by Corollary 3.6(i) that  $a_1^*, \ldots, a_\ell^*$  is  $[-g, \ldots, 0, 1, 2, \ldots, \ell - g]$ -regular.

If  $I^{\ell} \not\subseteq JI^{\ell-1}$  then choose  $k < \ell-1$  maximal such that  $I^{\ell} \subseteq JI^k$  and t minimal such that  $I^{\ell} \subseteq J_t I^k + I^{k+1}$ . Note that  $k \ge \ell - g$ . Let  $f = a_1 u_1 + \dots + a_t u_t + w \in$  $I^{\ell}$  where  $u_i \in I^k$  and  $w \in I^{k+1}$ . Then  $u_t^{*(k)} \in [(a_1^*, \dots, a_{t-1}^*) : a_t^*]_k =$  $(a_1^*, \dots, a_{t-1}^*)_k$ . Hence  $u_t \in J_{t-1}I^{k-1} + I^{k+1}$ . Thus  $f \in J_{t-1}I^k + I^{k+1}$ , contradicting our choice of k and t. Thus  $I^{\ell} \subseteq JI^{\ell-1}$  so  $r_J(I) \le \ell - 1$ .  $\Box$ 

LEMMA 8.3. Let (R, m) be a regular local ring containing a field and having infinite residue field. Let  $I \subseteq R$  be an ideal of analytic spread  $\ell$  such that G(I)satisfies  $(S_{\ell})$ . Let J be a minimal reduction. Then  $I^{\ell} \subseteq J(I \cap K_{\ell-1})$ . More generally, let  $a_1, \ldots, a_{\ell}$  be a basic generating set for J. If  $f = a_1u_1 + \cdots + a_nu_n \in I^2$ with  $u_i \in K_m$  with  $m \ge n-1$  then  $u_i \in I \cap K_m$ .

*Proof.* Since J is generated by a basic generating set we have  $n_i$  such that  $ht(J_iI^{n_i}: I^{n_i+1}) + I > i$  for  $g \leq i < \ell$ . Choose

$$x_{g+1} \in (I:K_m) \cap (J_g I^{n_g}:I^{n_g+1}) - \cup \{P \mid P \in Min^0(I)\}.$$

Now choose  $x_{i+1}$  inductively for i < n so that

$$x_{i+1} \in (I:K_m) \cap (J_i I^{n_i}:I^{n_i+1}) - \cup \{P \mid P \in \operatorname{Min}^0(I + (x_{g+1},\ldots,x_i))\}.$$

Under these conditions the ideal  $(a_1^*, \ldots, a_g^*, a_{g+1}^* + x_{g+1}^*, \ldots, a_j^* + x_j^*)$  has depth j on G(I) for  $j \leq \ell$  since G(I) satisfies  $(S_\ell)$ . Thus the Koszul complex on  $a_1^*, \ldots, a_j^* + x_j^*$  is acyclic.

Let f be as above. Then  $a_1^*u_1^{*(0)} + \cdots + a_n^*u_n^{*(0)} = 0$  in G(I). Note that  $x_i^*u_i^{*(0)} = 0$  for  $g + 1 \le i \le n$ . If the coefficients are not all in I then let  $i = \max\{t \mid u_t^{*(0)} \ne 0\}$ . Then

$$a_1^* u_1^{*(0)} + \dots + a_g^* u_g^{*(0)} + (a_{g+1}^* + x_{g+1}^*) u_{g+1}^{*(0)} + \dots + (a_i^* + x_i^*) u_i^{*(0)} = 0$$

and since G(I) satisfies  $(S_{\ell})$  we have  $u_i^{*(0)} \in (a_1^*, \ldots, a_g^*, a_{g+1}^* + x_{g+1}^*, \ldots, a_{i-1}^* + x_{i-1}^*)$ . Evaluating this sum in degree zero gives  $u_i \in (x_{g+1}, \ldots, x_i) + I$ . But the same analysis with  $x_{g+1}^n, \ldots, x_i^n$  in place of  $x_{g+1}, \ldots, x_i$  shows that  $u_i \in \bigcap_n((x_{g+1}^n, \ldots, x_i^n) + I) = I$ .  $\Box$ 

THEOREM 8.4. Let (R, m) be a regular local ring containing a field and having infinite residue field. Let  $I \subseteq R$  be an ideal with ad(I) = 1. If G(I) satisfies  $(S_{g+1})$ then  $r_J(I) \leq g$  for all minimal reductions  $J \subseteq I$ .

*Proof.* Let  $J = (a_1, \ldots, a_{g+1})$  be a minimal reduction of I generated by a basic generating set. If  $P \supseteq I$  and ht P < g + 1 then P is minimal over I. Hence  $r(I_P) < g$  since  $G(I_P)$  is CM.

Choose  $k \gg 0$  such that  $ht(J_g I^k : I^{k+1}) + I > g$  and let

$$x \in J_g : (J_g)^{\langle g+1 \rangle} \cap J_g I^k : I^{k+1} - \cup \{P \mid P \in \operatorname{Min}^0(I)\}.$$

Let  $f \in I^{g+1}$ . Suppose inductively that for some n < g,  $f = a_1v_1 + \cdots + a_{g+1}v_{g+1} \in JI^n$  where  $v_i \in I^n$  and  $xv_{g+1} \in J_g \cap I^n = J_gI^{n-1}$ . For n = 1 this holds by Lemma 8.3. Then  $(a_{g+1}^* + x^*)v_{g+1}^{*(n)} \in (a_1^*, \ldots, a_g^*)$ , so  $v_{g+1}^{*(n)} \in (a_1^*, \ldots, a_g^*)_n$ . Say  $v_{g+1} = a_1r_1 + \cdots + a_gr_g + w$  with  $w \in I^{n+1}$  and  $r_i \in I^{n-1}$ . Then  $xw \in J_q \cap I^{n+1} = J_qI^n$ . Also

 $f - a_{g+1}w \in J_g \cap I^{n+2} = J_g I^{n+1},$ 

so the inductive hypotheses are preserved. Hence  $I^{g+1} = JI^g$  and  $r_J(I) \leq g$ .  $\Box$ 

THEOREM 8.5. Let (R, m) be a regular local ring containing a field and having infinite residue field. Let  $I \subseteq R$  be an ideal of height g and ad(I) = 2. Suppose that  $r(I_P) \leq 1$  for every  $P \in Min^0(I)$ . If G(I) satisfies  $(S_{g+2})$  then  $r_J(I) \leq g+1$ for every minimal reduction  $J \subseteq I$ .

*Proof.* By Theorem 8.4 we have  $r(I_P) < \ell(I_P)$  for  $P \supseteq I$  and ht(P/I) < 2.

Let  $J = (a_1, \ldots, a_{g+2})$  be a minimal reduction of I generated by a basic generating set and choose  $k \gg 0$  such that  $ht(J_{g+1}I^k : I^{k+1}) + I > g + 1$ . Let

$$\begin{aligned} &x_1 \in J_g J_{g+1} : K_{g+1} \cap J_g : K_g \cap J_g I : I^2 - \cup \{P \mid P \in \operatorname{Min}^0(I)\}, \\ &x_2 \in J_{g+1} J_g^{\langle g+1 \rangle} : K_{g+1} \cap J_{g+1} I^k : I^{k+1} - \cup \{P \mid P \in \operatorname{Min}^0(I, x_1)\}. \end{aligned}$$

Let  $f \in I^{g+2}$  and say  $f = a_1u_1 + \dots + a_{g+2}u_{g+2}$  where  $u_i \in K_{g+1} \cap I$ . Then  $x_2^*u_{g+2}^{*(1)} = a_1^*t_1^{*(0)} + \dots + a_{g+1}^*t_{g+1}^{*(0)}$  where  $t_i \in J_g^{(g+1)}$ . Then since  $x_1^*u_i^{*(1)} = x_1^*t_i^{*(0)} = 0$  we have that  $(u_1^{*(1)} - t_1^{*(0)}, \dots, u_{g+1}^{*(1)} - t_{g+1}^{*(0)}, u_{g+2}^{*(1)})$  is a sum of Koszul relations on the regular sequence  $a_1^*, \dots, a_g^*, a_{g+1}^* + x_1^*, a_{g+2}^* + x_2^*$ . Write this vector as a sum of Koszul relations and let  $(y_0^{*(0)} + y_1^{*(1)} + \dots)$  be the coefficient of  $(0, \dots, 0, -(a_{g+2}^* + x_2^*), a_{g+1}^* + x_1^*)$ . We obtain the following information:

(1) 
$$u_{g+2}^{*(1)} - y_1^{*(1)} x_1^* - y_0^{*(0)} a_{g+1}^* \in (a_1^*, \dots, a_g^*),$$
  
(2)  $y_0^{*(0)} x_1^* = 0,$  and  
(3)  $y_1^{*(1)} a_{g+1}^* \in (a_1^*, \dots, a_g^*).$ 

From (1) and (2) we get  $(x_1^*)^2 y_1^{*(1)} \in (a_1^*, \dots, a_g^*)$ , which combined with (3) gives  $y_1^{*(1)} x_1^* \in (a_1^*, \dots, a_g^*) : (a_{g+1}^* + x_1^*) = (a_1^*, \dots, a_g^*)$ . Hence  $u_{g+2}^{*(1)} - y_0^{*(0)} a_{g+1}^* \in (a_1^*, \dots, a_g^*)$ . Write  $u_{g+2} = a_1r_1 + \dots + a_gr_g + y_0a_{g+1} + w$  with  $w \in I^2$ . Note that  $x_2w = a_1r_1' + a_gr_g' + a_{g+1}(t - x_2y_0)$  where  $t \in J_g^{\langle g+1 \rangle}$ . Then, since  $x_1(t - x_2y_0) \in I$  we get  $(t - x_2y_0)^{*(0)} \in (a_1^*, \dots, a_g^*)$ . Thus  $t - x_2y_0 \in I$ . We may conclude that  $x_2w \in J_{g+1}I$ .

Now,  $f - a_{g+2}w = a_1s_1 + \dots + a_gs_g + a_{g+1}(u_{g+1} + a_{g+2}y_0) \in I^3$ , where  $s_i \in I$ . Since  $x_1^*(u_{g+1}^{*(1)} + a_{g+2}^*y_0^{*(0)}) \in (a_1^*, \dots, a_g^*)$  we get  $u_{g+1} + a_{g+2}y_0 \in J_g + I^2$ , say  $u_{g+1} + a_{g+2}y_0 = a_1v_1 + \dots + a_gv_g + z$ . Thus  $f - a_{g+2}w - a_{g+1}z \in J_g \cap I^3 = J_gI^2$ , so  $f \in JI^2$ .

Suppose now that  $f \in I^{g+2} \cap JI^n$  for some n with  $2 \leq n < g+1$  and  $f = a_1r_1 + \cdots + a_gr_g + a_{g+1}z + a_{g+2}w$  where  $z, w \in I^n$  and  $x_2w \in J_{g+1}I^{n-1}$ . Note that  $x_1z \in J_gI^{n-1}$  by hypothesis. If  $x_2w = a_1t_1 + \cdots + a_{g+1}t_{g+1}$  with  $t_i \in I^{n-1}$  then we get that  $(\ldots, z^{*(n)} - t_{g+1}^{*(n-1)}, w^{*(n)})$  is a sum of Koszul relations on  $a_1^*, \ldots, a_g^*, a_{g+1}^* + x_1^*, a_{g+2}^* + x_2^*$ . As before write this as a sum of Koszul relations and let  $(\cdots + y_{n-1}^{*(n-1)} + y_n^{*(n)} + \cdots)$  be the coefficients of  $(0, \ldots, 0, -(a_{g+2}^* + x_2^*), a_{g+1}^* + x_1^*)$ . Then

(1) 
$$w^{*(n)} - x_1^* y_n^{*(n)} - a_{g+1}^* y_{n-1}^{*(n-1)} \in (a_1^*, \dots, a_g^*),$$
  
(2)  $x_1^* y_{n-1}^{*(n-1)} \in (a_1^*, \dots, a_g^*),$  and  
(3)  $y_n^{*(n)} a_{g+1}^* \in (a_1^*, \dots, a_g^*).$ 

From (1) and (2) we have  $(x_1^*)^2 y_n^{*(n)} \in (a_1^*, \ldots, a_g^*)$ , which combined with (3) gives  $y_n^{*(n)} x_1^* \in (a_1^*, \ldots, a_g^*)_n$ . Hence  $w^{*(n)} - a_{g+1}^* y_{n-1}^{*(n-1)} \in (a_1^*, \ldots, a_g^*)$ . Write  $w = a_1 r_1 + \cdots + a_g r_g + y_{n-1} a_{g+1} + w'$  with  $w' \in I^{n+1}$ . Then  $x_2 w' \in J_{g+1}I^{n-1} \cap I^{n+1}$ , say  $x_2 w' = a_1 r_1' + \cdots + a_{g+1} r_{g+1}'$ . Then since  $x_1 r_{g+1}' \in J_g I^{n-2}$  we get  $r_{g+1}' \in J_g I^{n-2} + I^n$ , so  $x_2 w' \in J_{g+1}I^n$ . Hence  $f - a_{g+2} w' =$ 

 $a_1s_1 + \dots + a_gs_g + a_{g+1}(z + a_{g+2}y_{n-1}) \in I^{n+2}$  with  $s_i \in I^n$ . Finally, since  $x_1(z + a_{g+2}y_{n-1}) \in J_gI^{n-1}$  we get  $z + a_{g+2}y_{n-1} \in J_gI^{n-1} + I^{n+1}$ . Hence  $f - a_{g+2}w' - a_{g+1}z' \in J_g \cap I^{n+2} = J_gI^{n+1}$ , so that  $f \in JI^{n+1}$ . Inductively, we now get  $I^{g+2} = JI^{g+1}$ .

#### References

- [AH] Aberbach I. and Huneke, C.: An improved Briançon–Skoda theorem with applications to the Cohen–Macaulayness of Rees rings, *Math. Ann.* 297 (1993) 343–369.
- [AHH] Aberbach, I. M., Huckaba, S. and Huneke, C.: Reduction numbers, Rees rings, and Pfaffian ideals, J. Pure Applied Alg. (to appear).
- [Ar] Artin, M.: Algebraic approximation of structures over complete local rings, *Publ. Math. I.H.E.S.* 36 (1969) 23–56.
- [AN] Artin, M. and Nagata, M.: Residual intersections in Cohen-Macaulay rings, J. Math. Kyoto Univ. 12 (1972) 307–323.
- [BaS] Bayer, D. and Stillman, M.: A criterion for detecting m-regularity, *Invent. Math.* 87 (1987) 1–11.
- [BS] Briançon, J. and Skoda, H.: Sur la clôture intégrale d'un idéal de germes de fonctions holomorphes en un point de  $C^n$ , C. R. Acad. Sci. Paris Sér A 278 (1974) 949–951.
- [Bu] Burch, L.: On ideals of finite homological dimension in local rings, *Proc. of Camb. Phil. Soc.* 64 (1968) 941–948.
- [CN] Cowsik, R. C. and Nori, M. V.: On the fibres of blowing up, J. of Indian Math. Soc. 40 (1976) 217–222.
- [CST] Cuong, N. T., Schenzel, P., and Trung, N. V.: Verallgemeinerte Cohen-Macaulay-Moduln, Math. Nachr. 85 (1978) 57-73.
- [EG] Eisenbud, D. and Goto, S.: Linear free resolutions and minimal multiplicities, J. Algebra 88 (1984) 89–133.
- [GH] Goto, S. and Huckaba, S.: On graded rings associated with analytic deviation one ideals, Amer. J. Math. (to appear).
- [GN1] Goto, S. and Nakamura, Y.: Cohen-Macaulay Rees algebras of ideals having analytic deviation two (preprint).
- [GN2] Goto, S. and Nakamura, Y.: Gorenstein graded rings associated to ideals of analytic deviation two (preprint).
- [GNi] Goto, S. and Nishida, K.: Filtrations and the Gorenstein property of the associated Rees algebras (preprint).
- [GS] Goto, S. and Shimoda, Y.: On the Rees algebras of Cohen–Macaulay local rings, *Lect. Notes in Pure and Applied Mathematics*, vol. 68, Marcel-Dekker, New York, 1979, pp. 201–231.
- [GW] Goto, S. and Watanabe, K.: On graded rings I, J. Math. Soc. Japan 30 (1978) 179-213.
- [GHO] Grothe, A., Herrmann, M., and Orbanz, U.: Graded rings associated to equimultiple ideals, Math. Z. 186 (1984) 531–566.
- [HI] Herrmann, M. and Ikeda, S.: On the Gorenstein property of Rees algebras, Manu. Math. 58 (1987) 471–490.
- [HHR] Herrmann, M., Huneke, C., and Ribbe, J.: On reduction exponents of ideals with Gorenstein form ring, *Proc. Edinburgh Math. Soc.* (to appear).
- [HRS] Herrmann, M., Ribbe, J., and Schenzel, P.: On the Gorenstein property of formrings (preprint).
- [HSV1] Herzog, J., Simis, A., and Vasconcelos, W. V.: On the arithmetic and homology of algebras of linear type, *Trans. Amer. Math. Soc.* 283 (1984) 661–683.

- [HSV2] Herzog, J., Simis, A., and Vasconcelos, W. V.: On the canonical module of the Rees algebra and the associated graded ring of an ideal, *J. of Alg.* 105 (1987) 285–302.
- [HoHu1] Hochster M. and Huneke, C.: tight closure, invariant theory, and the Briançon-Skoda theorem, J. Amer. Math. Soc. 3 (1990) 31-116.
- [HR] Hochster, M. and Ratliff, L. J., Jr.: Five theorems on Macaulay rings, *Pac. J. of Math.* 44 (1973) 147–172.
- [HH1] Huckaba, S. and Huneke, C.: Powers of ideals having small analytic deviation, Amer. J. Math. 114 (1992) 367–403.
- [HH2] Huckaba, S. and Huneke, C.: Rees algebras of ideals having small analytic deviation, *Trans. Amer. Math. Soc.* 339 (1993) 373–402.
- [HM] Huckaba, S. and Marley, T.: Depth properties of Rees algebras and associated graded rings, J. Algebra 156 (1993) 259–271.
- [Hu1] Huneke, C.: On the associated graded ring of an ideal, *Illinois J. Math.* 26 (1982) 121–137.
- [Hu2] Huneke, C.: The theory of d-sequences and powers of ideals, *Adv. in Math.* 46 (1982) 249–279.
- [I] Ikeda, S.: On the Gorenstein property of Rees algebras over local rings, Nagoya Math. J. 92 (1986) 135–154.
- [JK] Johnston, B. and Katz, D.: Castelnuovo regularity and graded rings associated to an ideal, *Proc. Amer. Math. Soc.* (to appear).
- [L] Lipman, J.: Cohen–Macaulayness in graded algebras, *Math. Res. Letters* 1 (1994) 149–157.
- [LS] Lipman, J. and Sathaye, A.: Jacobian ideals and a theorem of Briançon–Skoda, *Michigan Math. J.* 28 (1981) 199–222.
- [LT] Lipman, J. and Teissier, B.: Pseudo-rational local rings and a theorem of Briançon–Skoda about integral closures of ideals, *Michigan Math. J.* 28 (1981) 97–116.
- [MR] Matijevic, J. and Roberts, P.: A conjecture of Nagata on graded Cohen–Macaulay rings, J. Math. Kyoto Univ. 14 (1974) 125–138.
- [Ma] Matsumura, H.: Commutative Ring Theory, Cambridge University Press, Cambridge, 1986.
- [Mu] Mumford, D.: *Lectures on curves on an algebraic surface*, Princeton University Press, Princeton, New Jersey, 1966.
- [Mc] McAdam, S.: Asymptotic prime divisors, *Lect. Notes in Math. 1023*, Springer, 1983.
- [NR] Northcott, S. and Rees, D.: *Reductions of ideals in local rings, Math. Proc. Camb. Phil.* Soc. 50 (1954) 145–158.
- [NV] Noh, S. and Vasconcelos, W.: The S<sub>2</sub>-closure of a Rees algebra, *Results in Math.* 23 (1993) 149–162.
- [O] Oiishi, A.: Castelnuovo's regularity of graded rings and modules, *Hiroshima Math. J.* 12 (1982) 627–644.
- [Sa] Sally, J.: On the associated graded ring of a local Cohen–Macaulay ring, J. Math. Kyoto Univ. 17 (1977) 19–21.
- [SS] Sancho de Salas, J. B.: Géométrie Algébrique et applications I, La Rabida, Travaux en Cours, vol. 22, Hermann, Paris, 1987, pp. 201–209.
- [SUV] Simis, A., Ulrich, B., and Vasconcelos, W.: Cohen–Macaulay Rees algebras and degrees of polynomial relations, *Math. Annalen* (to appear).
- [SV] Simis, A. and Vasconcelos, W. V.: The syzygies of the conormal module, *Amer. J. of Math.* 103 (1981) 203–224.
- [Ta] Tang, Z.: Rees rings and associated graded rings of ideals having higher analytic deviation (preprint).
- [T1] Trung, N. V.: Reduction exponents and degree bound for the defining equations of graded rings, *Proc. Amer. Soc.* 101 (1987) 229–234.
- [T2] Trung, N. V.: Filter-regular sequences and multiplicity of blow-up rings of ideals of the principal class, *J. Math. Kyoto Univ.* (to appear).

I. M. ABERBACH ET A
---------------------

- [T3] Trung, N. V.: Reduction number, *a*-invariant, and Rees algebras of ideals having small analytic deviation, *Proceedings of the Workshop on Commutative Algebra*, Trieste 1992 (to appear).
- [TI] Trung, N. V. and Ikeda, S.: When is the Rees algebra Cohen–Macaulay?, *Comm. Algebra* 17(12) (1989) 2893–2922.
- [U] Ulrich, B.: Artin–Nagata properties and reductions of ideals, *Cont. Math.* 159 (1994) 373–400.
- [UV] Ulrich, B. and Vasconcelos, W. V.: The equations of Rees algebras of ideals with linear presentation, *Math. Z.* 214 (1993) 79–92.
- [VV] Valabrega, P. and Valla, G.: Form rings and regular sequences, *Nagoya Math. J.* 72 (1978) 91–101.
- [Va1] Vasconcelos, W. V.: On the equations of Rees algebras, J. reine angew. Math. 418 (1991) 189–218.
- [Va2] Vasconcelos, W.: Hilbert functions, analytic spread, and Koszul homology, *Cont. Math.* 159 (1994) 401–422.
- [Va3] Vasconcelos, W.: Arithmetic of Blowup Algebras (manuscript).
- [Vi] Viêt, D. Q.: A note on the Cohen-Macaulayness of Rees algebras of filtrations, *Comm. in Alg.* 21 (1993) 221–229.